NOVEMBER 1994

RESEARCH PAPER TWENTY-NINE

INDICES OF EFFECTIVE EXCHANGE RATES: A COMPARATIVE STUDY OF ETHIOPIA, KENYA AND THE SUDAN

ASMEROM KIDANE

ARCHIV 102042 NOMIC RESE

NOMIC RESEARCH CONSORTIUM

POUR LA RECHERCHE ECONOMIQUE EN AFRIQUE

IDRC - Lib. ID2042

Indices of effective exchange rates: a comparative study of Ethiopia, Kenya and the Sudan

This report is presented as received by IDRC from project recipient(s). It has not been subjected to peer review or other review processes.

This work is used with the permission of African Economic Research Consortium.

© 1994, African Economic Research Consortium.



anda 2000 - N 1007 1007 - **79** Other publications in the AERC Research Paper Series:

- Structural Adjustment Programmes and the Coffee Sector in Uganda by Germina Ssemogerere, Research Paper 1.
- Real Interest Rates and the Mobilization of Private Savings in Africa by F.M. Mwega, S.M. Ngola and N. Mwangi, Research Paper 2.
- Mobilizing Domestic Resources for Capital Formation in Ghana: The Role of Informal Financial Markets by Ernest Aryeetey and Fritz Gockel, Research Paper 3.
- The Informal Financial Sector and Macroeconomic Adjustment in Malawi by C. Chipeta and M.L.C. Mkandawire, Research Paper 4.
- The Effects of Non-Bank Financial Intermediaries on Demand for Money in Kenya by S.M. Ndele, Research Paper 5.
- Exchange Rate policy and Macroeconomic Performance in Ghana by C.D. Jebuni, N.K. Sowa and K.S. Tutu, Research Paper 6.
- A Macroeconomic-Demographic Model for Ethiopia by Asmerom Kidane, Research Paper 7.
- Macroeconomic Approach to External Debt: the Case of Nigeria by S. Ibi Ajayi, Research Paper 8.
- The Real Exchange Rate and Ghana's Agricultural Exports by K. Yerfi Fosu, Research Paper 9.
- The Relationship Between the Formal and Informal Sectors of the Financial Market in Ghana by E. Aryeetey, Research Paper 10.
- Financial System Regulation, Deregulation and Savings Mobilization in Nigeria by A. Soyibo and F. Adekanye, Research Paper 11.

The Savings-Investment Process in Nigeria: an Empirical Study of the Supply Side by A. Soyibo, Research Paper 12.

- Growth and Foreign Debt: the Ethiopian Experience, 1964-1986 by B. Degefe, Research Paper 13.
- Links Between the Informal and Formal/Semi-formal Financial Sectors in Malawi by C. Chipeta and M.L.C. Mkandawire, Research Paper 14.
- The Determinants of Fiscal Deficit and Fiscal Adjustment in Côte d'Ivoire by O. Kouassy and B. Bohoun, Research Paper 15.
- Small and Medium-Scale Enterprise Development in Nigeria by D.E. Ekpenyong and M.O. Nyong, Research Paper 16.
- The Nigerian Banking System in the Context of Policies of Financial Regulation and Deregulation by A. Soyibo and F. Adekanye, Research Paper 17.
- Scope, Structure and Policy Implications of Informal Financial Markets in Tanzania by M. Hyuha, M.O. Ndanshau and J.P. Kipokla, Research Paper 18.
- European Economic Integration and the Franc Zone: The Future of the CFA Franc after 1996 by A. M'Bet and M. Niamkey, Research Paper 19.
- Revenue Productivity Implications of Tax Reform in Tanzania by N. Osoro, Research Paper 20.
- The Informal and Semi-Formal Financial Sectors in Ethiopia: A Study of the Iqqub, Iddir and Savings and Credit Co-operatives by D. Aredo, Research Paper 21.
- Inflationary Trends and Control in Ghana by N.K. Sowa and J.K. Kwakye, Research Paper 22.
- Macroeconomic Constraints and Medium-Term Growth in Kenya: A Three-Gap Analysis by F.M. Mwega, N. Mwangi & K. Olewe-Ochilo, Research Paper 23.
- The foreign exchange market and the Dutch auction system in Ghana, by Cletus K. Dordunoo, Research Paper 24.

- Exchange rate depreciation and the structure of sectoral prices in Nigeria under an alternative pricing regime, 1986-89, by Olu Ajakaiye and Ode Ojowu, Research paper 25.
- Exchange rate depreciation, budget deficit and inflation the Nigerian experience, by Festus O. Egwaikhide, Louis N. Chete, and Gabriel O. Falokun, Research Paper 26.
- Trade, payments liberalization and economic performance in Ghana, by C.D. Jebuni, A.D. Oduro and K.A. Tutu, Research Paper 27.
- Constraints to the development and diversification of non-traditional exports in Uganda, 1981-90, by G.N. Ssemogerere and L.A. Kasekende, Research Paper 28.

Indices of effective exchange rates: a comparative study of Ethiopia, Kenya and the Sudan

Asmerom Kidane

University of Addis Ababa, Ethiopia

AERC Research Paper 29 African Economic Research Consortium, Nairobi November 1994 © African Economic Research Consortium, 1994

Typeset by Centre for the Study of African Economies, University of Oxford, St. Cross Building, Manor Road, Oxford, OX1 3UL, England.

Published by The African Economic Research Consortium, P.O. Box 62882, Nairobi, Kenya.

Printed by The Regal Press Kenya Ltd., P.O. Box 46166, Nairobi, Kenya.

ISBN 9966-900-16-0

Contents

List of tables List of figures Abstract

| I | Introduction | 1 |
|------|---|----|
| II | Exchange rate indices: nominal indices | 2 |
| III | Some conceptual and methodological issues | 9 |
| IV | The real effective exchange rate | 13 |
| V | Implications of the mathematical issues | 15 |
| VI | Relative price (cost) indices | 16 |
| VII | Relations between exchange rate indices and | |
| | macroeconomic performance | 19 |
| VIII | Empirical estimates of effective exchange rates | 21 |
| IX | Parallel market exchange rate | 37 |
| X | Conclusions | 52 |
| App | endix 1 | 53 |
| App | endix 2 | 54 |
| App | endix 3 | 55 |
| Note | s | 56 |
| Refe | rences | 57 |

List of tables

| Table 1. | Official exchange rates for Ethiopia, Kenya and the Sudan | 23 |
|-----------|--|----|
| Table 2. | Nominal effective exchange rates for Kenya, Ethiopia and | |
| | the Sudan, 1970=100 | 26 |
| Table 3. | Trend equations for nominal effective exchange rates for | |
| | Kenya, Ethiopia and the Sudan, 1970=100 | 30 |
| Table 4. | Nominal effective exchange rates for Kenya, Ethiopia and | |
| | the Sudan, 1980=100 | 31 |
| Table 5. | Trend equations for nominal effective exchange rates for | |
| | Kenya, Ethiopia and the Sudan, 1980=100 | 32 |
| Table 6. | Real effective exchange rates for Kenya, Ethiopia and the | |
| | Sudan, 1970=100 | 33 |
| Table 7. | Trend equations for real effective exchange rates for | |
| | Kenya, Ethiopia and the Sudan, 1970=100 | 34 |
| Table 8. | Real effective exchange rates for Kenya, Ethiopia and the | |
| | Sudan, 1980=100 | 39 |
| Table 9. | Trend equations for real effective exchange rates for | |
| | Kenya, Ethiopia and the Sudan, 1980=100 | 40 |
| Table 10. | Parallel exchange rates for Kenya, Ethiopia and the Sudan | 41 |
| Table 11. | Parallel effective exchange rates for Kenya, Ethiopia and | |
| | the Sudan, 1970=100 | 42 |
| Table 12. | Trend equations for parallel market nominal effective | |
| | exchange rate for Kenya, Ethiopia and the Sudan | 43 |
| Table 13. | Parallel nominal effective exchange rates for Kenya, | |
| | Ethiopia and the Sudan | 44 |
| Table 14. | Trend equations for parallel market nominal effective exchange | |
| | rates for Kenya, Ethiopia and the Sudan, 1980=100 | 45 |
| Table 15. | Parallel real effective exchange rates for Kenya, | |
| | Ethiopia and the Sudan, 1970=100 | 46 |

| Table 16. | Trend equations for parallel market real effective | |
|-----------|---|----|
| | exchange rates for Kenya, Ethiopia and the Sudan, 1970=100 | 47 |
| Table 17. | Parallel real effective exchange rates for Kenya, | |
| | Ethiopia and the Sudan, 1980=100 | 48 |
| Table 18. | Trend equations for parallel market real effective exchange | |
| | rates for Kenya, Ethiopia and the Sudan, 1980=100 | 49 |

List of figures

| Figure 1. | Nominal effective exchange rates for Kenya | 27 |
|-----------|---|----|
| Figure 2. | Nominal effecitve exchange rates for Ethiopia | 28 |
| Figure 3. | Nominal effective exchange rates for the Sudan | 28 |
| Figure 4. | Real effective exchange rates for Kenya | 35 |
| Figure 5. | Real effective exchange rates for Ethiopia | 35 |
| Figure 6. | Real effective exchange rates for the Sudan | 36 |
| Figure 7. | Parallel nominal effective exchange rates for Kenya | 50 |
| Figure 8. | Parallel nominal effective exchange rates for Ethiopia | 50 |
| Figure 9. | Parallel nominal effective exchange rates for the Sudan | 51 |

Abstract

The paper considers the various indices of effective exchange rate that are applied in many countries to measure the overvaluation or undervaluation of a particular currency compared to the currency of major trading countries. First the conceptual issues of the nominal effective exchange rate (NEER) is considered. In general there are three types of nominal effective exchange rates namely the export weighted, import weighted and trade weighted rates. Other indices may also be developed on the basis of the three indices. The major drawback with these rates is that they do not isolate the effect of overvaluation from possible inflationary differentials between reporting countries and major trading partners. In order to isolate the pure exchange rate effect, the nominal effective exchange rate should be deflated by the ratio of the inflation rate of a reporting country to that of a partner country. This would in turn give us the Real Effective Exchange Rate (REER). There are two problems associated with the conversion of NEER to REER. First, there is an issue of what type of price index to use. There are several indices including the Consumer Price Index (CPI), the wholesale price index (WPI), as well as other related indices. Second, even if a particular index is chosen, that index may not be measured in a similar manner between the two countries. If the Real Effective Exchange Rate (REER) is measured with minimal error then such an index may be a measure of changes in the price of tradables compared to non-tradables. In other words, the REER is akin to the Real Exchange Rate (RER).

I Introduction

This paper looks at the adequacy and consistency of exchange rate indices for African countries. Such indices have been estimated for many developed countries since the early 1970s. The indices are adequate for comparing exchange rate movements between industrialized countries and with comparable economic performances. When one tries to estimate indices of effective exchange rate between a developing country on the one hand and between major trading countries (which are economically more developed) then the adequacy of such indices may come into question. In this paper we will examine the various indices, what they are expected to measure and whether they are adequate for developing countries. Balassa (1987) examines two definitions of the real exchange rate; the traditional and the modern. The traditional approach estimates the nominal effective exchange rate and then deflates it by relative price movement of a given and a partner country, so as to achieve finally the 'pure' exchange rate indicator. The latter is referred to as the real effective exchange rate. The modern version estimates the ratio of price indices of traded and non-traded goods. In the modern version, a simple model is formulated whereby export demand (through foreign and domestic export supply) equations are estimated. A similar formulation may be developed for imports. This is commonly referred to as a multilateral effective exchange rate (MERM). The traditional method seems to be rather simplistic, in other words such indices do not seem to take into account the theoretical macroeconomic issues into explaining the movement of a particular country's exchange rate vis-a-vis the major trading partners. Thus a modern and more comprehensive method of estimating and effective exhange rate is developed.¹

In this study we shall only consider the traditional definition of the exchange rate. Section II reviews various indices of nominal and real effective exchange rate and Section III considers some of the conceptual and methodological issues involved in constructing such indices. Sections IV to VI briefly discuss the real effective exchange rate. Section VII identifies those characteristics of developing countries which may have an effect on exchange rate movements. Sections VIII and IX discuss the results of nominal, real and parallel market exchange rates, while Section X gives some concluding remarks.

Il Exchange rate indices: nominal indices

Most African currencies are pegged against one of the major currencies. For example the Ethiopian birr is pegged against the US dollar, while the Kenyan shilling and Sudanese sterling are pegged against the British pound. However, in a world where currency floating is the order of the day, a fixed exchange rate is inappropriate. This raises the question as to the exchange rate of an African currency vis à vis the major trading partners of the country under consideration. In order to estimate the exchange rate of an African currency at a given time, one needs to identify that country's major trading partners. Once the major trading partners have been identified the next step is to devise an index that expresses an "average" change in the exchange rate of a particular African currency against the currencies of its major trading partners. The simplest way of doing this would be to calculate the simple arithmetical mean of individual bilateral exchange rates; however, this approach is valid only if each trading partner has the same share of trade with the particular country under study. This is obviously not the case.

The most reasonable approach would be to have a weighted "average" of bilateral exchange rates. The latter method is what is known as an effective exchange rate. Hirsch and Higgins (1970) were among the first to develop this concept of effective exchange rate. They recognised that, for analytical purposes, the exchange rate of a particular currency should "reflect the evolving relationship between that currency and all other currencies". They also identified that the effective exchange rate should be an index and should not be expressed in absolute terms, and that only the currencies of major trading partners (not all the currencies) should be included.

Development of an effective exchange rate

After recognizing that the effective exchange rate should be an index and be expressed in relative terms, Hirsch and Higgins defined an effective exchange rate of any given currency as "the percentage 'direct' change in the numeraire rate

INDICES OF EFFECTIVE EXCHANGE RATES

minus the weighted percentage 'indirect' change in the numeraire rates of other currencies". Thus, given n currencies (1,2,...,n) with the currency n as the numeraire, N is the percentage change in the numeraire rate while $w_1, w_2, ..., w_n$ are the relevant weights. The percentage change in the effective exchange rate E, can be expressed as:

(1)
$$E' = n' - \sum_{i=2}^{n} W_i N_i$$

Alternatively, Equation 1 may be rewritten as the ratio of the numeraire rate of a given currency to the sum of weighted changes in numeraire rates of other currencies. In other words,

(2)
$$E' = n' / \sum_{i=2}^{n} W_i N_i$$

To further clarify the indices in Equations 1 and 2, given a particular currency where the dollar is the numeraire, the exchange rate of that particular currency is equal to the percentage change in its dollar rates of other currencies. In Equation 1, N^1 is known as the direct effect while $\sum W_i N_i$ is the indirect effect. If the numeraire rates of other currencies show zero change, the indirect effect will be insignificant. After Equation 1 or 2 has been estimated for each period a base period is chosen and an index is formulated.

Rhomberg (1976) elaborates more on the construction of indices of exchange rate and discusses issues that have to be taken into consideration when constructing effective exchange rates. Like Hirsch and Higgins he confines his analysis to developed western economies. In other words, the unique conditions and characteristics of developing countries are not taken into consideration.

When considering the choice of a base period in the construction of an index, Rhomberg states that the base date should be chosen in such a way that the period is as close to the equilibrium rates as possible.

According to Rhomberg, weighting is an important issue in constructing an effective exchange rate index. The choice of weights is said to depend on the objective of the study or on the economic analysis. One may have as many weights (indices) as the number of policy issues that one wishes to analyze. In Rhomberg's words,

The proper choice of weights depends, therefore, on the particular policy objective selected as focal point of the index. For different objectives, indices employing different weighting schemes would be appropriate.... (Rhomberg, 1976, p.89).

The most commonly considered issue is the study of the effect of changes in the exchange rate on a given country's balance of payments. In this case, the proper choice of weight could be an average of bilateral export and import weights. If one is to study the effect of exchange rate changes on the cost of imports, the appropriate weight would be the value of imports from major trading partners. This is because the change in the price of foreign currency in terms of the domestic currency based on bilateral import weight measures the effect of a set of foreign exchange changes on the cost of imports in local currency. If one wishes to study the effect of exchange rate changes on the competitiveness of a given country's exports in the world market, the appropriate weight would then be the volume of exports. The reason behind this weighting system is that the change of the domestic currency in terms of currencies of trading partners weighted by the volume (share) of exports would help indicate the cost of the home country's exports to foreign customers. One could give other examples showing the correspondence between a particular policy issue and an appropriate weighting system.

Rhomberg discusses seven indices that are commonly used by various financial institutions. The author states that it is generally difficult to compare the various indices because they may differ on the choice of the sample countries included, the base period and the data inputs. Even though Rhomberg considers seven different indices, there are essentially four, while the other remaining three are simply extensions.

The import weighted index

The import weighted index is defined as the average price of partner currencies in terms of domestic currency relative to the base period, the weights being the partners' share in total imports of the home country. Symbolically this may be written as

(3)
$$E M_i = \sum_j \frac{M_{ji}}{\sum_j M_{ji}} R_{ji}$$

where

 M_{ji} = imports of country *i* from country *j*

 $\Sigma \dot{M}_{ii}$ = total imports of county *i*

 $1/R_{ji} = R_{ji}$ = price of one unit of currency *i* in terms of currency *j* expressed as an index relative to the base period (summation is over all the currency of major trading partners).

The bilateral export weighted index

The bilateral export weighted index is defined as the arithmetical mean of the price of the domestic currency in terms of partner currencies, relative to the base period, and weighted by the partners' share in the total export of the home country. Symbolically, it is expressed as follows:

(4)
$$EX = \sum \left(\frac{X_{ij}}{\sum X_{ij}}\right)R_{ij}$$

where

 X_{ij} = exports of country *i* to country *j* X_{ij} = total exports of country *i*

The bilateral trade weighted index

The bilateral trade weighted index is the arithmetical mean of Equation 4 and reciprocal of Equation 3, weighted by the ratio of exports and imports to their sum. This is symbolically expressed as

(5)
$$EMX = \frac{\left[M_{i}\left(\frac{1}{EM_{i}}\right) + X_{i}(EX_{i})\right]}{M_{i} + X_{xi}}$$

where

 M_{i} = total imports of country *i* X_{i} = total exports of country *i* to the world market.

The global export weighted index

The global export weighted index is defined as the arithmetical mean of prices of domestic currency in terms of foreign currencies, relative to the base period, and weighted by foreign countries' share in total exports to the world market. The market of the home country is not included. This may be written as

(6)
$$EXW_i \ge \sum_{i \neq j} \left[\frac{(X_{i,} - X_{ij})}{\sum_{i \neq j} (X_{i,} - X_{ij})} \right] R_{ij}$$

where $(X_{i}, -X_{i}) = \text{total export of country } i \text{ minus export of } i \text{ to country } j$.

The average export weighted index

The average export weighted index is a simple arithmetical mean of the bilateral export weighted index and the global export weighted index and is expressed as

(7) EXBW = 0.5 EX + 0.5 EMX

The average trade weighted index

The average trade weighted index is a simple arithmetical mean of the average export weighted index and the reciprocal of the import weighted index. This may be expressed as

(8)
$$EMXBW = 0.5 \left[\frac{1}{EM_i}\right] + 0.5 [EXBW]$$

The last index discussed by Rhomberg is based on the more complex MERM and is not considered here.

Uses of the various indices

The import weighted index EM_i would help studies of the effect of changes in the exchange rate on the cost of imports, while the bilateral export weighted index EX_i may help analysts understand the effect of changes in the exchange rate on the competitiveness of a given country's exports.

The bilateral trade weighted index (*EMX*), which is an average of two indices weighted by bilateral export and import weights, is used as a measure of an overall trade movement. One of the indices (*EM*_i) is inverted because the export and import weighted indices move in opposite directions as the index is the weighted average of changes in the price of foreign currencies in terms of domestic currency. The other indices are weighted averages of prices of domestic currency in terms of foreign currency. The average trade weighted index (*EMXBW*), which incorporates import weights as well as the two types of export weights, can be regarded as a trade-weighted index. This formula is expected to show the change in one structure of trade as a result of changes in the exchange rate. This formula is also expected to give some indication about price responses to exchange rate changes. This index is closer to the MERM type indices which use the general equilibrium approach to obtain an appropriate weight.

Rhomberg states that the best method of measuring indices is to use the MERM system whereby a general equilibrium model is developed and the effect of exchange rate change on several economic variables could be studied. The author uses the seven indices to compare movements of effective exchange rates for the developed countries of Western Europe. In order to make a valid comparison, the author chooses the same base period and the same set of exchange rate data. The indices showed general conformity with respect to upward and downward movement over the study period. The MERM indices showed trends that are neither too high nor too low and the author concludes that the MERM weighted index is the better measure.

III Some conceptual and methodological issues

The indices of effective exchange rate, as suggested by Hirsch and Higgins (1970), Rhomberg (1976) and others, does not seem to consider some major methodological and conceptual issues involved in the construction and interpretation of the various indices. More specifically, the informational content of different indices and their interpretation in relation to different market conditions needs to be analyzed. Moreover, the indices already summarized do not seem to differentiate between the nominal and real effective exchange rate. Maciejewski (1983) states that a meaningful interpretation of the various nominal and real exchange rates depends on the combination of four issues, namely the proper choice of a base period; the proper choice of weighting and the policy question being addressed; the plausibility of the relative price or cost indicator; and the mathematical formulation. The author then develops the concepts of nominal and real effective exchange rates and states that an index can only answer one question at a time. More specifically, the index answers the question of the effects of changes in exchange rate relative to the base period. It is confined to a given set of currencies.

One major issue that Maciejewski addresses in appraising various indices is the need to link economic theory as related to exchange rate movements and the process of index construction. Again this link can only be realized by considering one issue at a time. For example, if one wants to assess the impact of changes in the exchange rate on the "competitiveness of exports", the appropriate weight could be the volume of exports and a Laspeyre-type index where constant quantity and constant quality baskets of goods are assumed. The standard Laspeyre price index that is commonly used in the construction of wholesale, retail, cost of living and other indices is given by

(9)
$$P_{La}^{01} = \frac{\sum P_1 Q_0}{\sum P_0 Q_0}$$

where P_0 and P_1 are base and current period prices Q_0 is the weight and is nothing but the base period quantity purchased.

By the same logic one may develop a Laspeyre-type exchange rate index. The bilateral export weighted index of Equation 4 may be rewritten so that it looks similar to Equation 9. This is done as follows:

(10) NEER =
$$\frac{\sum_{j} X_{0}^{p} \left(\frac{e_{1}^{p}}{E_{1}^{r}}\right)}{\sum_{j} X_{0}^{p} \left(\frac{e_{0}^{p}}{e_{0}^{r}}\right)}$$

where

- X^p = value of export to the *p*th trading partner in the base period (the value is expressed in partner country's currency);
- e_1^p = dollar value (dollar over currency) of one unit of *p*th partner in period 1;
- e_0^p = dollar value of one unit of *p*th partner country in the base period;
- e_1^r = dollar value of one unit of reporting country's (country under study) currency in period 1;
- $e_0^{\rm r}$ = dollar value of one unit of reporting country's currency in the base period;
- NEER = nominal effective exchange rate.

It may easily be noted that X_0^p in Equation 10 plays the role of Q_0 in Equation 9, while e_0^p / e_0^r and e_1^p / e_1^r play the role of P_0 and P_1 in the same equation. A simple manipulation of Equation 10 yields the following results:

(11)
$$E = \sum_{p} \frac{W_{0}^{p}}{\sum W_{0}^{p}} \left(\frac{e_{1}^{p}}{e_{0}^{p}} - \frac{e_{1}^{r}}{e_{0}^{r}} \right)$$

where

$$W_0^p = X_0^p \frac{e_0^p}{e_0^r}.$$

One should note that Equation 11 is merely a bilateral export weighted index where

| W_{0}^{p} / w_{0}^{p} | = weight; |
|-----------------------------|--|
| e_1^p / e_0^p | = indices of change from base period exchange rate in P 's |
| | exchange rate; |
| $e_1^{\rm r} / e_0^{\rm r}$ | = indices of change from base period in reporting country's |
| | numeraire exchange rate. |

Again, Equation 11 is similar to Equation 9 where

$$W_0^p = X_{ij}$$
$$R_{ij} = \frac{e_1^p}{e_0^p} - \frac{e_1}{e_0}$$

The above formula explains how the value of a country's earnings from commodity exports in the current period, with the value of its earnings in the base period, changes as a result of cumulative effects of exchange rate movements. Contrary to the common view, it is not a direct measure of international competitiveness in a given country's export sector because competitiveness is a function of many different economic variables.

It is also possible to change the weights of Equation 11 to find out the effect of exchange rate movement on export earnings. For example, if one wants to study the effect of changes in exchange rate on a country's export earning from a group

of specific commodities, then the weight ought to be the distribution among the principal competitors of specific market shares in the commodities in the base period.

Implications of using Laspeyre-type indices

We have already stated that Laspeyre-type indices of effective exchange rate assume an unchanged product composition as well as unchanged quality of goods. In other words, the above indices require a changing exchange rate policy so as to maintain the *status quo* of the base period. This approach may be unrealistic in a world where the quantity and quality of products and services change. At the same time, developing countries are facing a changing world with deteriorating terms of trade in the export of primary commodities such as coffee and cocoa. The use of Laspeyre-type exchange rate indices within the context of developing countries may thus be questionable.

The second problem with Laspeyre-type exchange rate indices is that such indices are biased downwards compared with other indices such as Paasche. The latter uses the current period as a base. The extent of this downward bias has not been verified, but chances are that they are likely to be substantial.

The preceding paragraphs show that the bilateral export weighted index is similar to a Laspeyre-type index and briefly considered the consequences of using such indices. Similar discussions can be developed for Equations 3 and 6. Equations 3, 4 and 6 are usually referred to as the nominal effective exchange rates. We will now briefly consider the real effective exchange rate. The discussion will continue to be confined to the bilateral export weighted exchange rate.

IV The real effective exchange rate

The nominal effective exchange rate (NEER) is not adjusted or deflated for a corresponding change in relative prices. If the NEER is applied to study the effect of exchange rate policy, then the "real" exchange rate effect may not be isolated. This is because the NEER may also include the effect of inflation differentials between a country under consideration and its partners. Thus, in order to isolate the pure exchange rate effect on export "performance", the NEER may have to be deflated so as to arrive at the EER. Unlike the NEER, the real effective exchange rate (REER) may not be used to measure the effect of undervaluation or overvaluation of a currency.² In this particular case, the REER may be interpreted as a gain or loss in international price competitiveness between the current and the base period.

Below we present the corresponding value for the REER for the bilateral export weighted index. If we deflate Equation 4 by a ratio of price changes from the base period in partner countries to the country under consideration, we will get the following result

(12)
$$REER = \sum_{p} Z_{p} \left(\frac{e_{1}^{pr}}{e_{0}^{pr}} \right) \left(\frac{p_{1}^{pr}}{p_{0}^{pr}} \right)$$

where

- $Z_p = W_0^p / W_0^p$ = bilateral share of export; $e_0^{pr} = e_0^p / e_0^p$ = bilateral exchange rate of partner countries to home country in the base period;
- $e_0^{\rm pr} = e_0^{\rm p} / e_1^{\rm r}$ = bilateral exchange rate of partner countries to home currency in the current period;
- $p_0^{\rm pr} = p_0^{\rm p} / p_0^{\rm r}$ = ratio of export prices of partner countries to home country in the base period;

 $p_1^{\text{pr}} = p_1^{\text{p}} / p_1^{\text{r}}$ = ratio of export price of partner countries relative to home country in the current period.

Equation 12 is the NEER the same as Equation 4 but with deflated price differentials between reporting and partner countries. Equation 12 can be written more simply as:

$$(13) \quad R = Z_p \left(\frac{r_1^p}{r_0^p} \right)$$

where

$$r_1^P = \frac{e_1^{pr}}{p_1^{pr}}$$

and

$$r_0^P = \frac{e_0^{Pr}}{e_0^{Pr}}$$

V Implications of the mathematical formulations

Each of the indices described above includes a relative price or cost component. The method of averaging and the calculation of the relative prices will have an effect on the indices finally calculated. To estimate the real effective exchange rate we may easily estimate two separate indices relative price and exchange rate changes. The same weighting procedure will be used for each and they will then be combined. This approach enables one to decompose the variation in real exchange rate variability into exchange rate change and price change components.

Exchange rate changes are generally easy to obtain or estimate but this is not so for price changes. To find these, one may have to use proxies for price change indices and combine them with the exchange rate indices. This approach is obviously more restrictive and the interpretation of the final indices may not be easy.

The method of averaging may also affect the calculated value of such indices. The arithmetical mean, the geometrical mean and the harmonic mean are most commonly used. Each of these averaging methods has its own biases; however, the geometrical mean seems to work best because it treats appreciating or depreciating values in a symmetrical manner and is not affected by extreme values. In other words, the averaging process is a linear function of the components of bilateral exchange rate indices.

VI Relative price (cost) indices

We have already stated that the NEER should be adjusted by relative price (cost) indices in order to eliminate inflation differentials and to obtain the pure exchange rate change effects. There are different relative price indices that are useful for deflating NEER indices. These indices are widely used in the industrial countries of the West. Below, we consider the most important ones.

Index of export unit values

The index of export unit values compares the value of exports of the home country with those of its competitors. The indices are based on proxies where export unit value is the proxy for prices. They may be easily estimated provided that one obtains data on value and quantities. While it is easy to estimate the unit values, the use of the latter as a proxy for price of exports has been questioned. This is because the changes in the proxy value may be due not only to changes in prices but also to compositional shifts in the selected set of commodities.

Furthermore, indices of export unit values may not be regarded as indices of price competitiveness because export values also include different taxes and export prices may not be increased or decreased in response to competitiveness. In general, the index of export unit values may be regarded as a indicator of price competitiveness of manufactured goods.

Index of wholesale price

The index of wholesale price compares the wholesale prices of the home country to those of competitors. The indices are expected to reflect price developments for "potentially exportable" products. However, they do include non-traded goods as well as imported goods. They may thus incorporate indirect taxes or subsidies

INDICES OF EFFECTIVE EXCHANGE RATES

levied on imports. Because of these factors, wholesale price indices may overstate or understate the changes in export prices. In addition, wholesale price indices may temporarily not be comparable because of differences in scope and coverage of the regimes as well as the method of averaging. It may be possible to derive a wholesale price index for the purpose of deflating changes in exchange rates.

Cost indices

Cost indices compare labour costs of the home country to those of competitors and are a measure of profitability of exports. Labour is engaged in both tradable and non-tradable goods and may not be a good indicator of cost-based competitiveness in international trade. The labour cost is normally defined as the ratio of labour costs to the volume of output produced, and when this index is taken as a measure of costs it assumes that labour costs are a major input in the process of production or that other costs are similar across all competing countries. Another problem of the labour-based cost index is that it may incorporate erratic movements and may not give a true indication of cost trends. Such erratic movements may be due to a "lead-lag" relation.

Consumer price indices

Consumer price indices (CPI) measure the consumer price of reporting country vis à vis competitors and are used as a proxy for measuring relative total unit producer costs. If the CPI goes up it could imply that the relative cost of production will also go up. This may eventually lead to reduced profits in tradables and reduce supply and market shares. If a CPI is used for measuring producers costs, we are assuming that CPIs are a major determinant of wages as well as other inputs. The problem with the CPI is that it is concerned with intercountry comparisons and is thus flawed when one compares labour-intensive and capital-intensive structures of production, as is the case when one compares a developed country with a less developed country.

The GDP deflator index

The GDP deflator index measures a home country's GDP deflators with the competitors' GDP deflators and is a measure of total unit costs of a country and its partners. If properly measured, it may give a satisfactory indication of the profitability of exported goods of a reporting country against its competitors.

VII Relation between exchange rate indices and macroeconomic performances

Even though the aim here is to empirically estimate indices of the effective exchange rates and interpret the results vis a vis the performance in enhancing export earnings, reducing the cost of imports and measuring the international competitiveness of the three countries, it may be in order to briefly consider the macroeconomic structure from which some of the propositions about the relationship between indices and outcomes can be derived.

The three countries under study have faced serious economic crises, over the past 15 years. These crises have manifested themselves in various forms such as low growth rate of GDP, increasing balance of payments deficits, high debt and debt servicing followed by accumulation of arrears. The causes for this poor economic performance are many and varied. Some of these are deteriorating terms of trade, high interest rate and protectionism. By far the most important cause could be overall macroeconomic mismanagement. This is again manifested by lack of appropriate incentive schemes which failed to promote efficiency in the utilization of resources.

One of the consequences of poor macroeconomic management is the overvaluation of a country's currency which discourages export and diversification and makes imports artificially cheap. The quantitative restrictions that are introduced in the allocation of foreign exchange are the result of an overvalued domestic currency. As a result of an overvalued exchange rate a country always faces balance of payment deficits. This is because overvalued exchange rate makes production for exports unprofitable while imports become cheaper. Few individuals or institutions will have access to exchange rates at an artificially low price and privileged importers earn high profits. On the other hand manufactured exports are produced at high cost and this leads to a loss in international competitiveness. Domestic demand for tradables will be high because of low relative prices and lack of competitiveness.

Devaluation of the existing currency is expected to narrow the gap in the current account deficit by increasing the profitability of exports and making the latter more competitive. At the same time, cost of imports will be increasing. The effectiveness of the policy depends on the elasticity of export supply and the demand for imports. On the other hand devaluation may have a short-term inflationary effect. For example, the cost of production of agricultural produce is likely to increase as a result of devaluation and is passed on to consumers. The inflationary effect could be substantial if the fraction of imported inputs is high. Also an overvalued exchange rate implies an implicit tax on exports. Since this implicit tax is not used as a budgetary support but as subsidy to importers, the overvalued currency is likely to lead to a budgetary deficit. Also, an overvalued exchange rate will lead to the emergence of a parallel market. Devaluation is expected to reduce or eliminate the premium and realign the rates. It is with this type of macroeconomic management that we have to estimate various indices of effective exchange rate. One may be able to compare the empirical estimate of each country's indices with the consequences that may emerge.

VIII Empirical estimates of effective exchange rates

In this and subsequent sections, empirical estimates of various exchange rates will be presented for Kenya, Ethiopia and the Sudan.

The choice of these three countries is based on several considerations. First, the required data for each country is available. Second, even though the three countries are within the same region and are classified among the least developed countries, there is substantial variation between them. Out of the three, Kenya has relatively high per capita income; export earning is from more diversified commodities and services, GDP from the industrial sector constitutes a substantial proportion and the latter is more modern with little excess capacity. The government of Kenya is committed to the functioning of a market orientated, relatively free and open economy. Most important the exchange rate policy has been relatively more flexible, exchange controls are less rigid and the list of prohibited import items is not long. Because of this the parallel market is quite thin and the premium very low. On the other extreme, Ethiopia has one of the lowest per capita incomes in the world, more than 80% of the export earnings are from a single primary commodity (coffee), and the industrial sector is both inefficient and highly protected with a large percentage of excess capacity. The reason for the latter is the high dependency on imported capital and intermediate goods. Until 1991 the government had been pursuing orthodox Marxist economic policy whereby most of the major industrial and service establishments have been under state control. In general the incentive structure that is a prerequisite for conducive private investment did not exist. The exchange rate policy has been rigid. Because of declining export earnings and very low aid and capital inflow, the foreign exchange resource is insignificant resulting in strict foreign exchange control with a long list of prohibited import items. Allocation of the available meagre foreign exchange was centralized; in other words it was not based on supply and demand. The official exchange rate of the local currency that has been pegged to a single currency has been constant throughout. As a result the parallel market began to flourish with a premium reaching as much as 362%.

Compared with these two countries the economic structure of the Sudan is somewhere between the two. Even though the mainstay of the economy is agriculture, the latter seem to be capital intensive and depend on imported inputs; the same may be said of the industrial sector. Like Ethiopia, an appropriate incentive structure does not seem to exist. Past attempts at structural adjustments in general and attempted exchange rate realignment could not be sustained and did not produce the desired result. Even though there were a series of devaluations, the currency was still overvalued and strict exchange control prevails. There is also an ever increasing parallel market, however, the premium was not as high as that of Ethiopia.

In general the choice of the three countries would enable us to construct indices of effective exchange rate for developing economies with different macroeconomic environments.

For each of these three countries, an index of EER movements has been constructed in accordance with the formulae given in Equations 3, 4 and 12. The indices that have been constructed are the nominal and the real effective exchange rates, for both the official and the parallel markets, for each of the three countries. The EERs include:

- export weighed NEER;
- import weighted NEER;
- trade weighted NEER;
- export weighted REER;
- import weighted REER; and
- trade weighted REER.

The estimates were made by assuming two different base periods, 1970 and 1980. In all, we have estimated 24 different indices for each of the three countries. Trend equations will also be estimated for each series.³ Before going on to examine the empirical results of the various exchange rates, we will consider the official exchange rates of the three countries. Table 1 shows that there is very little variability in the official exchange rate of Ethiopia, while there is substantial variability in Kenya's official exchange rate. Sudan lies somewhere between the two.

| | | | | |
|----------|----------|-------|-------|--|
| Year | Ethiopia | Kenya | Sudan | |
| 1970 | 2.50 | 7.14 | 2.87 | |
| 1971 | 2.50 | 7.14 | 2.87 | |
| 1972 | 2.50 | 7.14 | 2.87 | |
| 1973 | 2.07 | 6.90 | 2.87 | |
| 1974 | 2.07 | 7.14 | 2.87 | |
| 1975 | 2.07 | 8.26 | 2.87 | |
| 1976 | 2.07 | 8.31 | 2.87 | |
| 1977 | 2.07 | 7.95 | 2.87 | |
| 1978 | 2.07 | 7.40 | 2.50 | |
| 1979 | 2.07 | 7.33 | 2.00 | |
| 1980 | 2.07 | 7.33 | 0.50 | |
| 1981 | 2.07 | 7.57 | 0.60 | |
| 1982 | 2.07 | 10.21 | 1.00 | |
| 1983 | 2.07 | 12.72 | 1.30 | |
| 1984 | 2.07 | 13.79 | 1.30 | |
| 1985 | 2.07 | 15.78 | 2 30 | |
| 1986 | 2 07 | 16.28 | 2 50 | |
| 1987 | 2.07 | 16.04 | 3.00 | |
| 1988 | 2.07 | 17.70 | 0.00 | |
| | | | | |

Table 1 Official exchange rates for Ethiopia, Kenya and the Sudan

Source: International Monetary Fund, Financial Statistics (various issues). Note: values are expressed in terms of one US dollar.

Some preliminary indicators

A simplified table showing the relationship between export with major trading partners and the corresponding bilateral exchange rate is given for 1970 and 1980 in Appendix 1. In 1970 Ethiopian exports were predominantly to the United States, while the other six partner countries took only 23% of the total. By 1980 the proportion of Ethiopian exports to the US had decreased dramatically while those to Japan, France, Germany and Italy showed a relative increase. This may be explained by the change in the exchange rate of the Ethiopian birr relative to the trading partners. Between the two periods, the Ethiopian birr appreciated by
about 17% relative to the US dollar. On the other hand, the Ethiopian birr depreciated by 71.4, 44.3 and 54.1% against the currencies of Japan, France and Germany.

The distribution of Kenya's exports to the major trading partners is less skewed than for the other two countries. The United Kingdom appears to be the major receiver of Kenya's exports, while between 1970 and 1980, Kenya's exports to Japan, France, Germany and Italy increased. This could be explained by the large percentage depreciation of the Kenyan shilling relative to the currencies of Kenya's trading partners. The Sudan's exports to major trading partners shows a highly skewed distribution in 1980 compared with 1970. The United Kingdom is the most important buyer of Sudan's exports. Unlike Ethiopia and Kenya the variation in the Sudan's exports could not be explained by changes in bilateral exchange rates.

We have also tried to relate the change in imports of 1970 and 1980 to bilateral exchange rate (Appendix 2). In relating changes in imports by partner country to bilateral exchange rate, the latter is measured as the ratio of the partner country's currency over the home currency. With respect to Ethiopia one observes that Japan, Germany, and Italy were the three major sources of Ethiopian imports. There is not much change in the share of imports between 1970 and 1980. On the other hand, the Ethiopian currency shows some appreciation between the two periods suggesting that the change in imports could not be explained by the change in the corresponding bilateral exchange rates.

The distribution of sources of Kenyan imports seems to be similar to that of exports. Britain is the main source even though its share had decreased by 8% between 1970 and 1980. Imports from Germany, Italy and the Netherlands increased considerably in 1980 and this may be explained by changes in the bilateral exchange rates.

The Sudan's imports from its major trading partners shows a less skewed distribution in 1980 compared with 1970, with the United Kingdom becoming a relatively less important source of imports. The increase in imports from Germany and Italy may be explained by the relative appreciation of the Sudanese currency.

The preceding discussion suggests that variation in exports and imports to and from major trading partners may be partially explained by changes in bilateral exchange rate. This is particularly so for Kenya. Given that the data in Appendices 1 and 2 give some indication of the possible relationship between exports, imports and exchange rates, the next step would be to apply the indices of EER, as explained previously, so as to see the patterns of such indices and to give economic interpretation to our results.

Results and economic interpretations

Nominal effective exchange rates

We have already stated that one can develop as many indices as there are policy issues to be answered. In our empirical estimation we have confined ourselves to nominal export weighted and nominal import weighted indices, as well as the average between the two. The empirical results of each of the three indices should give some indication of the effects of changes in exchange rate on the competitiveness of a given country's exports and effects of the cost of imports on the extent of trade movements.

Taking 1970 as a base period, the three indices of the three countries are given in Table 2 as well as in Figures 1, 2 and 3. The export and import weights used in estimating the effective exchange rates are the values given in Appendices 1 and 2. Trend equations of the three indices are also given in Table 3. An increase in the trend value of an export weighted nominal exchange rate implies that a given country's currency has been depreciated over the study period. This is because a bilateral exchange rate is defined as dollar value of a reporting country's currency over dollar value of partner country's currency. The results clearly show that the Kenyan currency has been continuously depreciated; the extent of depreciation seems to be high during the 1980s compared with the 1970s. The result suggests that Kenya's exportables are likely to be competitive and export earning could show some increase. The opposite seems to be true for Ethiopia. The effective exchange rate in Figure 2, as well as the trend equation in Table 3, show that the Ethiopian currency is relatively overvalued, suggesting that the main cause of poor export performance for Ethiopia is the appreciation of its currency compared with those of its major trading partners. Also during the study period the Sudanese currency showed some relative appreciation.

The import weighted indices of the three countries are estimated by defining the bilateral exchange rates as the ratio of those of the partner countries to those of the reporting countries (Kenya, Ethiopia and the Sudan). Because of this the import weighted EER should move in the opposite direction to the export weighted EER. For purpose of comparison the import weighted indices have been inverted. Based on this interpretation, Kenya's cost of imports seems to be lower than that for Ethiopia and the Sudan. Finally, the trade weighted NEER is given in column 3 of Table 2. The results suggest that Kenya has experienced extensive trade movements, while this is not so for Ethiopia and the Sudan.

| | - 100 | , | | | | | | | | |
|------|-------|----------|------|------|----------|------|------|-------|------|---|
| | | Kenya | | E | Ithiopia | | | Sudan | | - |
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 | |
| 1970 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | - |
| 1971 | 0.98 | 1.04 | 1.04 | 1.01 | 1.01 | 1.01 | 1.01 | 0.97 | 0.99 | |
| 1972 | 1.03 | 1.05 | 1.04 | 1.01 | 1.04 | 1.02 | 1.05 | 1.03 | 1.04 | |
| 1973 | 1.03 | 1.13 | 1.09 | 0.84 | 0.90 | 0.87 | 1.11 | 1.06 | 1.08 | |
| 1974 | 1.08 | 1.29 | 1.19 | 0.84 | 0.97 | 0.91 | 1.16 | 1.08 | 1.12 | |
| 1975 | 1.30 | 1.43 | 1.38 | 0.84 | 0.98 | 0.91 | 1.17 | 1.18 | 1.17 | |
| 1976 | 1.50 | 1.58 | 1.53 | 0.85 | 1.03 | 0.94 | 1.26 | 1.35 | 1.31 | |
| 1977 | 1.40 | 1.59 | 1.48 | 0.86 | 1.08 | 0.97 | 1.27 | 1.27 | 1.27 | |
| 1978 | 1.28 | 1.66 | 1.47 | 0.87 | 1.17 | 1.02 | 1.16 | 1.09 | 1.13 | |
| 1979 | 1.23 | 1.68 | 1.46 | 0.87 | 1.02 | 1.02 | 0.93 | 0.84 | 0.88 | |
| 1980 | 1.19 | 1.58 | 1.38 | 0.86 | 1.02 | 0.98 | 0.86 | 0.77 | 0.82 | |
| 1981 | 1.87 | 1.55 | 1.71 | 0.87 | 1.29 | 1.08 | 0.58 | 0.78 | 0.68 | |
| 1982 | 1.87 | 1.85 | 1.86 | 0.86 | 1.11 | 0.98 | 0.34 | 0.38 | 0.36 | |
| 1983 | 2.43 | 2.08 | 2.25 | 0.86 | 1.10 | 0.97 | 0.34 | 0.40 | 0.36 | |
| 1984 | 2.98 | 2.11 | 2.54 | 0.85 | 1.10 | 0.97 | 0.18 | 0.24 | 0.21 | |
| 1985 | 3.07 | 2.79 | 2.93 | 0.87 | 1.11 | 0.99 | 0.18 | 0.21 | 0.20 | |
| 1986 | 3.26 | 3.47 | 3.36 | 0.87 | 1.22 | 1.05 | 0.11 | 0.12 | 0.12 | |
| 1987 | 2.96 | 3.93 | 3.45 | 0.89 | 1.11 | 1.10 | 0.12 | 0.11 | 0.11 | |
| 1988 | 3.23 | 4.33 | 3.79 | 0.88 | 1.23 | 1.05 | 0.11 | 0.10 | 0.10 | |
| Х | 1.86 | 1.95 | 1.89 | 0.88 | 1.10 | 6.99 | 0.73 | 0.74 | 0.73 | |
| S | 0.87 | 0.98 | 0.90 | 0.05 | 0.11 | 0.06 | 0.45 | 0.43 | 0.44 | |
| | | | | | | | | | | |

Table 2Nominal effective exchange rate for Kenya, Ethiopia and the Sudan, 1970= 100

Source: author's calculation, based on Table 1. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

We also tried to see the movement of the three indices when the base period was changed from 1970 to 1980. The results in Appendices I and II show a major change in the export and import weights of the three countries.

The NEER of the three countries is given in Tables 4 and 5. There seems to be little difference in the export weighted, import weighted or trade weighted indices of Ethiopia and the Sudan when 1980 wight is applied. On the other hand,

changing the base to 1980 has reduced the extent of Kenya's currency depreciation. In other words, by changing the base to 1980, Kenya's competitiveness in exports is lower, the cost of its imports is relatively higher and the extent of trade movements seems to have been reduced.

The results in the tables, as well as the estimated trend equations, can be compared in different ways and one may provide many different interpretations by making several assumptions. When we compare the results in Table 2 with the hypothesized consequences of a misaligned exchange rate as well as the outcomes of the selected but relevant macroeconomic indicators, the relations seems to show some consistency. The relevant macroeconomic indicators for the three countries is given in Appendix 3. The results show significant differences between Kenya on the one hand and Ethiopia and the Sudan on the other. First, the volume of Kenya's exports in is much higher than that of Ethiopia and the Sudan; second, the ratio of imports to exports is lower in Kenya compared with Ethiopia. The rate of inflation seems to be lower in Ethiopia compared with Kenya and the Sudan. The latter statement suggests the expected relation between devaluation and inflation. Even though a properly formulated and empirically estimated macroeconomic model is required to see the impact of devaluation on key macroeconomic variables, our approach, though simple, seems to verify the negative consequences of overvalued exchange rates.

In Section IV we stated that the NEER may not explain export performance, import cost or trade movement unless the index is deflated by a ratio in price of a partner and a reporting country. However, if the nominal exchange rate is deflated so as to give the pure exchange rate effect, then this deflated EER will help us to measure the trend in international price competitiveness between the base and current period.





Figure 2 Nominal effective exchange rates for Ethiopia, 1970=100







INDICES OF EFFECTIVE EXCHANGE RATES

Taking 1970 as a base, the export weighted, import weighted and trade weighted indices are given in Tables 6 and 7 as well as Figures 4, 5 and 6. For the Sudan and Ethiopia there seems to be little difference between the nominal and REER, suggesting that the ability to compete in international trade seems to be low due to overvalued currency. Similar results may also be observed when the base period is changed to 1980. By contrast, Kenya seems to have a relatively more competitive position in international trade even though the extent of competitiveness is reduced when the base period is changed from 1970 to 1980. Maciejweski states that under specific assumptions the real effective exchange rate is akin to the relative price ratio between tradables and non-tradables. The results in Table 6 show a relatively higher price of tradables to non-tradables in Kenya; this does not seem to be the case for Ethiopia and the Sudan.

| Table 3 | Trend equations for r | nominal | effective | exchange ra | tes for Ker | iya, Ethi | opia and th | e Sudan, 197 | 0=100 | |
|-------------|-----------------------|---------|-----------|-------------|-------------|-----------|-------------|--------------|-------|-------|
| | | | Kenya | | | Ethiopia | | | Sudan | |
| Model | Coefficients | - | ึง | ო | - | 5 | ε | | 2 | en l |
| Linear | Intercept | 1.42 | 0.42 | 0.42 | 0.93 | 0.95 | 0.94 | 1.44 | 1.40 | 1.42 |
| | | 2.43 | 1.85 | 2.45 | 40.89 | 27.75 | 38.32 | 13.28 | 13.22 | 13.41 |
| | Slope | 0.14 | 0.16 | 0.15 | -0.10 | 0.02 | 0.01 | -0.07 | -0.06 | -0.07 |
| | | 9.17 | 7.63 | 9.67 | -2.42 | 5.02 | 2.36 | -7.418 | -7.16 | -7.34 |
| | ۳ ۲ | 0.82 | 0.77 | 0.84 | 0.21 | 0.57 | 0.28 | 0.75 | 0.74 | 0.75 |
| | S.E. | 0.36 | 0.48 | 0.36 | 0.05 | 0.07 | 0.05 | 0.27 | 0.22 | 0.22 |
| Exponential | Intercept | 0.14 | 0.13 | 0.14 | 0.98 | 0.91 | 0.94 | 1.53 | 1.48 | 1.50 |
| | - | 0.37 | 0.02 | 0.35 | 38.24 | 17.27 | 26.07 | 6.95 | 6.95 | 6.44 |
| | Slope | 9.81 | 0.88 | 0.85 | -0.05 | 0.09 | 0.02 | -0.38 | -0.35 | -0.37 |
| | - - | 4.83 | 4.34 | 4.86 | -4.16 | 4.00 | 1.43 | -3.85 | -3.74 | -3.82 |
| | ۳ ۲ | 0.55 | 0.50 | 0.56 | 0.48 | 0.46 | 0.05 | 0.44 | 0.42 | 0.44 |
| | S.E. | 0.58 | 0.69 | 0.60 | 0.04 | 0.08 | 0.06 | 0.34 | 0.33 | 0.33 |
| | | | | | | | | | | |
| | | | | | | | | | | |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted

INDICES OF EFFECTIVE EXCHANGE RATES

Table 4

Nominal effective exchange rates for Kenya, Ethiopia and the Sudan, 1980 = 100

| | | Kenya | | E | thiopia | | | Sudan | |
|------|------|-------|------|------|---------|------|------|-------|------|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 |
| 1970 | 0.74 | 0.81 | 0.77 | 1.03 | 0.97 | 0.99 | 1.32 | 1.20 | 1.26 |
| 1971 | 0.75 | 0.80 | 0.78 | 1.04 | 0.97 | 1.01 | 0.27 | 1.20 | 1.24 |
| 1972 | 0.78 | 0.83 | 0.81 | 1.06 | 0.99 | 1.03 | 1.36 | 1.26 | 1.31 |
| 1973 | 0.88 | 0.84 | 0.82 | 0.90 | 0.86 | 0.88 | 1.40 | 1.32 | 1.36 |
| 1974 | 0.87 | 0.90 | 0.88 | 0.93 | 0.90 | 0.92 | 1.41 | 1.37 | 1.39 |
| 1975 | 1.03 | 1.08 | 1.05 | 0.93 | 0.92 | 0.93 | 1.57 | 1.39 | 1.48 |
| 1976 | 1.14 | 1.19 | 1.17 | 0.95 | 0.96 | 0.96 | 1.81 | 1.49 | 1.65 |
| 1977 | 1.09 | 1.12 | 1.11 | 0.99 | 0.99 | 0.99 | 1.68 | 1.49 | 1.59 |
| 1978 | 1.06 | 1.06 | 1.06 | 1.03 | 1.06 | 1.04 | 1.43 | 1.35 | 1.39 |
| 1979 | 1.05 | 1.04 | 1.05 | 1.04 | 1.05 | 1.05 | 1.09 | 1.07 | 1.08 |
| 1980 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| 1981 | 1.32 | 1.42 | 1.37 | 1.03 | 1.18 | 1.01 | 1.07 | 0.70 | 0.89 |
| 1982 | 1.41 | 1.49 | 1.45 | 0.97 | 1.02 | 0.99 | 0.50 | 0.40 | 0.45 |
| 1983 | 1.75 | 1.88 | 1.82 | 0.95 | 1.01 | 0.98 | 0.54 | 0.39 | 0.47 |
| 1984 | 2.03 | 2.42 | 2.14 | 0.93 | 1.01 | 0.98 | 0.34 | 0.22 | 0.28 |
| 1985 | 2.26 | 2.40 | 2.33 | 0.96 | 1.03 | 0.99 | 0.29 | 0.21 | 0.25 |
| 1986 | 2.52 | 2.60 | 2.56 | 1.02 | 1.11 | 1.06 | 0.16 | 0.13 | 0.14 |
| 1987 | 2.51 | 2.47 | 2.49 | 1.08 | 1.17 | 1.12 | 0.14 | 0.13 | 0.13 |
| 1988 | 2.67 | 2.69 | 2.68 | 1.04 | 1.10 | 1.08 | 0.13 | 0.12 | 0.13 |
| Х | 1.41 | 1.46 | 1.43 | 0.99 | 1.01 | 0.98 | 0.86 | 0.93 | 0.93 |
| S | 0.66 | 0.68 | 0.67 | 0.05 | 0.08 | 0.57 | 0.53 | 0.55 | |

Source: author's calculation.

Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| | | |) | | | • | | | |
|--------------|---|---|---|--|--|---|--|--|-------|
| | | Kenya | | ш | thiopia | | - | Sudan | |
| Coefficients | - | 5 | З | - | 5 | e | - | 5 | e |
| Intercept | 0.32 | 0.35 | 0.34 | 0.98 | 0.91 | 0.95 | 1.85 | 1.70 | 1.78 |
| - | 2.59 | 2.68 | 2.66 | 38.55 | 32.56 | 36.94 | 13.04 | 13.75 | 13.57 |
| Slope | 0.11 | 0.11 | 0.11 | 0.01 | 0.01 | 0.01 | -0.09 | -0.08 | -0.09 |
| - | 10.10 | 9.63 | 9.92 | 0.06 | 4.23 | 2.62 | -7.03 | -7.73 | -7.46 |
| Ъ, | 0.85 | 0.84 | 0.85 | -0.04 | 0.50 | 0.25 | 0.73 | 0.77 | 0.75 |
| S.E. | 0.26 | 0.28 | 0.27 | 0.05 | 0.06 | 0.05 | 0.30 | 0.26 | 0.27 |
| Intercept | 0.10 | 0.13 | 0.16 | 1.00 | 0.89 | 0.94 | 1.95 | 1.82 | 1.89 |
| | 0.35 | 0.45 | 0.41 | 28.7 | 20.53 | 24.96 | 6.99 | 7.09 | 7.04 |
| Slope | 0.63 | 0.64 | 0.64 | -0.01 | 0.06 | 0.03 | -0.47 | -0.46 | -0.47 |
| | 5.00 | 4.89 | 4.96 | -1.20 | 3.18 | 1.76 | -3.76 | -3.98 | 3.85 |
| ۲ ۲ | 0.57 | 0.56 | 0.57 | -0.06 | 0.33 | 0.11 | 0.41 | 0.45 | 0.43 |
| S.E. | 0.43 | 0.45 | 0.44 | 0.05 | 0.07 | 0.06 | 0.44 | 0.40 | 0.41 |
| | | | | | | | | | |
| | Coefficients Intercept Slope S.E. Slope R ² S.E. | Coefficients 1 Intercept 0.32 Slope 0.11 R ² 0.85 S.E. 0.85 Slope 0.10 Intercept 0.10 R ² 0.63 Slope 5.00 R ² 0.63 S.E. 0.43 | Coefficients 1 2 Intercept 0.32 0.35 Intercept 0.32 0.35 Slope 0.11 0.11 Slope 0.11 0.11 R ² 0.85 0.84 S.E. 0.26 0.28 Intercept 0.10 0.13 Intercept 0.10 0.13 Slope 0.64 0.45 Slope 0.63 0.64 S.E. 0.50 4.89 F ² 0.45 0.56 S.E. 0.43 0.56 | Coefficients 1 2 3 Intercept 0.32 0.35 0.34 Intercept 0.32 0.35 0.34 Slope 0.11 0.11 0.11 Slope 0.11 0.11 0.11 R ² 0.85 0.84 0.85 R ² 0.26 0.28 0.41 Slope 0.10 9.63 9.92 R ² 0.26 0.28 0.45 0.41 Slope 0.10 9.63 9.92 0.41 Siope 0.56 0.27 0.45 0.41 Slope 0.45 0.45 0.41 0.55 Size 0.43 0.45 0.45 0.55 Size 0.43 0.45 0.55 0.57 | Coefficients 1 2 3 1 E Intercept 0.32 0.35 0.34 0.98 38.55 1 Intercept 0.11 0.11 0.11 0.11 0.11 0.01 Slope 0.11 0.11 0.11 0.11 0.11 0.01 R ² 0.85 0.84 0.85 0.04 0.06 0.06 R ² 0.26 0.28 0.04 0.85 0.04 0.06 Sibpe 0.10 0.13 0.16 1.00 0.06 0.06 R ² 0.45 0.45 0.41 28.7 0.01 Sibpe 0.63 0.64 0.64 0.01 28.7 Sibpe 0.45 0.45 0.44 0.01 28.7 Sibpe 0.45 0.44 0.66 -1.20 0.06 R ² 0.45 0.45 0.06 -1.20 0.06 Sibpe 0.45 0.44 0.06 | Coefficients 1 2 3 1 Ethiopia Intercept 0.32 0.35 0.34 0.98 0.91 Intercept 0.32 0.35 0.34 0.98 0.91 Slope 0.11 0.11 0.11 0.11 0.01 0.01 Slope 0.11 0.11 0.11 0.11 0.01 0.01 0.01 Slope 0.11 0.11 0.11 0.11 0.01 0.01 0.01 Slope 0.10 9.63 9.92 0.06 4.23 Tracept 0.26 0.28 0.27 0.05 0.06 Intercept 0.10 0.16 0.16 0.05 0.06 4.23 Slope 0.50 4.496 -0.01 0.06 0.318 0.65 0.066 0.318 Slope 0.45 0.45 0.05 0.066 0.318 0.318 F. 0.45 0.65 0.066 0.33 0 | $ \begin{array}{c cccc} \mbox{Coefficients} & 1 & \mbox{Kenya} \\ \mbox{Intercept} & 1 & \mbox{Kenya} \\ \mbox{Intercept} & 0.32 & 0.35 & 0.34 & 0.98 & 0.91 & 0.95 \\ \mbox{Slope} & 0.11 & 0.11 & 0.11 & 0.01 & 0.01 & 0.01 & 0.01 \\ \mbox{Slope} & 0.11 & 0.11 & 0.11 & 0.01 & 0.01 & 0.01 & 0.01 \\ \mbox{Slope} & 0.26 & 0.28 & 2.66 & 38.55 & 32.56 & 36.94 \\ \mbox{Slope} & 0.10 & 9.63 & 9.92 & 0.06 & 4.23 & 2.62 \\ \mbox{Intercept} & 0.10 & 0.13 & 0.16 & 4.23 & 2.62 \\ \mbox{Intercept} & 0.10 & 0.13 & 0.16 & 1.00 & 0.89 & 0.94 \\ \mbox{Intercept} & 0.35 & 0.45 & 0.41 & 28.7 & 2053 & 24.96 \\ \mbox{Slope} & 0.63 & 0.64 & 0.64 & -0.01 & 0.06 & 0.03 \\ \mbox{Slope} & 0.65 & 0.57 & 0.66 & 0.03 & 0.11 \\ \mbox{F}^2 & 0.45 & 0.44 & -0.01 & 0.06 & 0.03 \\ \mbox{Slope} & 0.63 & 0.45 & 0.44 & -0.01 & 0.06 & 0.03 \\ \mbox{Slope} & 0.63 & 0.64 & 0.64 & -1.20 & 3.18 & 1.76 \\ \mbox{F}^2 & 0.43 & 0.45 & 0.44 & -0.01 & 0.05 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & -0.01 & 0.05 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & -0.01 & 0.05 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & 0.05 & 0.07 & 0.06 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & 0.05 & 0.07 & 0.06 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & 0.05 & 0.07 & 0.06 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.43 & 0.45 & 0.44 & 0.05 & 0.07 & 0.06 & 0.03 & 0.11 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.06 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.05 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.05 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.05 & 0.07 & 0.05 & 0.07 & 0.05 \\ \mbox{S.E.} & 0.44 & 0.05 & 0.07 & 0.07 & 0.07 & 0.07 & 0.07 & 0.05 \\ \mbox{S.E.} & 0.44 & 0.05 &$ | $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$ | |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| | | Kenya | | 1 | Ethiopia | | | Sudan | | |
|------|------|-------|------|------|----------|------|------|-------|------|--|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 | |
| 1970 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | |
| 1971 | 1.09 | 1.09 | 1.09 | 1.34 | 1.04 | 1.02 | 1.06 | 1.02 | 1.04 | |
| 1972 | 1.24 | 1.30 | 1.27 | 1.16 | 1.16 | 1.15 | 1.00 | 1.00 | 1.00 | |
| 1973 | 1.03 | 1.09 | 1.06 | 1.29 | 1.20 | 1.18 | 0.99 | 1.00 | 1.00 | |
| 1974 | 1.09 | 1.16 | 1.13 | 1.42 | 1.27 | 1.23 | 0.94 | 0.98 | 0.96 | |
| 1975 | 1.31 | 1.46 | 1.36 | 1.66 | 1.47 | 1.44 | 0.84 | 0.95 | 0.89 | |
| 1976 | 1.31 | 1.55 | 1.43 | 1.50 | 1.71 | 1.68 | 0.89 | 0.97 | 0.93 | |
| 1977 | 1.19 | 1.39 | 1.29 | 1.35 | 1.57 | 1.53 | 0.86 | 0.96 | 0.91 | |
| 1978 | 1.08 | 1.27 | 1.18 | 1.34 | 1.45 | 1.40 | 0.76 | 0.93 | 0.84 | |
| 1979 | 0.99 | 1.15 | 1.07 | 1.32 | 1.44 | 1.39 | 0.59 | 0.87 | 0.73 | |
| 1980 | 1.03 | 1.17 | 1.10 | 2.16 | 1.39 | 1.36 | 0.52 | 0.84 | 0.68 | |
| 1981 | 1.96 | 2.51 | 2.24 | 1.79 | 2.11 | 2.13 | 0.36 | 0.78 | 0.57 | |
| 1982 | 1.40 | 1.69 | 1.55 | 2.12 | 1.77 | 1.78 | 0.31 | 0.77 | 0.54 | |
| 1983 | 1.57 | 1.92 | 1.74 | 5.56 | 2.05 | 2.09 | 0.29 | 0.76 | 0.52 | |
| 1984 | 1.80 | 2.25 | 2.02 | 2.43 | 2.43 | 2.50 | 0.25 | 0.75 | 0.50 | |
| 1985 | 1.58 | 1.94 | 1.76 | 2.41 | 2.31 | 2.37 | 0.25 | 0.75 | 0.50 | |
| 1986 | 1.58 | 1.94 | 1.76 | 2.05 | 2.34 | 2.38 | 0.24 | 0.74 | 0.49 | |
| 1987 | 1.43 | 1.72 | 1.57 | 2.49 | 2.14 | 2.09 | 0.24 | 0.75 | 0.49 | |
| 1988 | 1.42 | 1.71 | 1.57 | | 2.60 | 2.54 | 0.24 | 0.74 | 0.49 | |
| x | 1.32 | 1.54 | 1.43 | 1.69 | 1.71 | 1.70 | 0.61 | 0.87 | 0.74 | |
| S | 0.20 | 0.42 | 0.35 | 0.54 | 0.50 | 0.52 | 0.33 | 0.11 | 0.22 | |

| Table (| 5 Rea | l effective | exchange | rates for | Kenya, | Ethiop | ia and th | e Sudan |
|---------|-------|-------------|----------|-----------|--------|--------|-----------|---------|
| | | | <u> </u> | | | | | |

Source: author's calculation.

Notes: 1 = Export weighted; 2 = Import weighted; 3 = Trade weighted.

| Table 7 | Trend equations for re | eal effec | tive exc | hange rates fo | or Kenya, | Ethiopia | and the Suc | lan | | |
|-------------|------------------------|-----------|----------|----------------|-----------|----------|-------------|--------|--------|--------|
| | | _ | Kenya | | ш | thiopia | | | Sudan | |
| Model | Coefficients | - | ึ่ง | ო | - | 2 | e | - | 5 | e |
| Linear | Intercept | 0.82 | 0.88 | 0.85 | 1.00 | 1.00 | 1.00 | 1.18 | 1.06 | 1.12 |
| | - | 7.29 | 9.10 | 8.40 | 9.47 | 6.83 | 7.75 | 27.14 | 69.13 | 38.17 |
| | Slope | 0.09 | 0.08 | 0.08 | 0.03 | 0.05 | 0.04 | -0.06 | -0.01 | -0.04 |
| | | 8.76 | 10.41 | 9.58 | 3.48 | 4.22 | 3.92 | -14.79 | -14.07 | -14.61 |
| | \mathbb{R}^2 | 0.81 | 0.86 | 0.83 | 0.38 | 0.48 | 0.44 | 0.92 | 0.91 | 0.92 |
| | S.E. | 0.23 | 0.19 | 0.21 | 0.22 | 0.31 | 0.26 | 0.09 | 0.03 | 0.06 |
| Exponential | Intercept | 0.56 | 0.61 | 0.58 | 0.89 | 0.79 | 0.84 | 1.25 | 1.21 | 1.23 |
| - | | 2.71 | 3.45 | 3.07 | 3.94 | 3.84 | 4.73 | 12.86 | 30.31 | 17.49 |
| | Slope | 0.55 | 0.53 | 0.54 | 0.21 | 0.36 | 0.28 | -0.36 | -0.12 | -0.24 |
| | - | 5.86 | 6.40 | 6.26 | 3.10 | 3.90 | 3.57 | -7.51 | -7.15 | -7.41 |
| | \mathbb{R}^2 | 0.65 | 0.71 | 0.68 | 0.32 | 0.44 | 0.40 | 0.75 | 0.74 | 0.75 |
| | S.E. | 0.31 | 0.27 | 0.29 | 0.23 | 0.32 | 0.27 | 0.16 | 0.06 | 0.11 |
| | | | | | | | | | | |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.









Figure 6 Real effective exchange rates for the Sudan, 1970=100



IX Parallel market exchange rate

In economies where the exchange rate is overvalued there are strict official exchange rate controls. The government resorts to rationing by classifying imported goods as "essentials" or "non-essentials". Restrictions are usually imposed on the latter. Even if one is allowed to import essential commodities, there is a highly bureaucratic application procedure.

These controls lead to corruption within the official government channels and this eventually leads to the development of parallel markets where the foreign currency (usually the US dollar) is sold at a much higher rate than the official rate. Visitors and exporters prefer to use the parallel market. Remittances sent by residents in Western countries are transferred through nonbank intermediaries and this reduces the amount of foreign exchange passing through to national banks or other authorized dealers. The parallel market exchange rate fluctuates according to the rules of demand and supply for the dollar or other currency. In economies where overvalued currency has been in existence for a long period, the parallel market for foreign exchange provides a larger portion of financing for imports.

It is argued that the parallel market exchange rate is an appropriate rate since the parallel market is more or less a floating exchange rate, the value of which is governed by supply and demand. Most nominal and real EERs are estimated using the official exchange rates. In economies with a large parallel market the official nominal and real effective exchange may not show the true picture. One may have also to estimate EER based on the parallel market. If it is possible to estimate the share of the parallel market, then a weighted average of the official and parallel market effective exchange rate would give a better indication of the extent of appreciation or depreciation of a currency. Unfortunately this is not easy to ascertain and time-series data on parallel exchange rates are not easy to obtain.

However, an estimate of parallel market exchange rates was made available by this author covering Kenya, Ethiopia and the Sudan. These results are given in Table 10. As expected, the difference between the official exchange rates and those of the parallel market seem to be low for Kenya and high for Ethiopia. For Ethiopia, the difference between the two rates seems to have Ethiopia. For Ethiopia, the difference between the two rates seems to have widen in the late 1980s compared with the early 1970s. For the sudan, there is no observable pattern.

Having assembled the estimated parallel market exchange rate we then derived the nominal and real EERs based on parallel markets. The results are given in Table 11 and 18.

When the NEERs based on official rates are compared with those based on parallel market rates, the Ethiopian case seems to show that in the parallel market the extent of overvaluation is much reduced, while for Kenya the opposite seems to be the case. There does not seem to be much difference between the two rates for the Sudan. The trend equations for the parallel exchange rates do not show a trend. Real effective exchange rates have also been estimated for the parallel market. The results for Kenya and the Sudan show little difference between the indices based on the official rates and those based on parallel rates (Tables 15 17). This does not seem to be the case for Ethiopia. In addition, the trend equations for parallel market-based real exchange rates for Ethiopia and the Sudan seem to be highly irregular (Tables 16 18).

INDICES OF EFFECTIVE EXCHANGE RATES

| _ | | _ |
|----|---|---|
| Та | h | 0 |
| 10 | U | 0 |

Real effective exchange rates for Kenya, Ethiopia and the Sudan, 1980 = 100

| •••••• | | Kenya | | E | Ethiopia | | | Sudan | |
|--------|------|-------|------|------|----------|------|------|-------|------|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 |
| 1970 | 0.92 | 0.85 | 0.89 | 0.73 | 0.75 | 0.74 | 1.33 | 1.88 | 1.66 |
| 1971 | 1.03 | 0.93 | 0.98 | 0.77 | 0.77 | 0.77 | 1.37 | 1.98 | 1.63 |
| 1972 | 1.18 | 1.05 | 1.14 | 0.86 | 0.86 | 0.86 | 1.32 | 1.89 | 1.61 |
| 1973 | 1.01 | 0.92 | 1.97 | 0.89 | 0.87 | 0.88 | 1.32 | 1.87 | 1.59 |
| 1974 | 1.09 | 0.99 | 1.03 | 0.93 | 0.90 | 0.92 | 1.28 | 1.78 | 1.53 |
| 1975 | 1.25 | 1.25 | 1.25 | 1.07 | 1.07 | 1.07 | 1.21 | 1.60 | 1.40 |
| 1976 | 1.25 | 1.33 | 1.29 | 1.23 | 1.25 | 1.24 | 1.22 | 1.61 | 1.42 |
| 1977 | 1.09 | 1.08 | 1.09 | 1.04 | 1.02 | 1.03 | 1.16 | 1.43 | 1.29 |
| 1978 | 1.09 | 1.08 | 1.08 | 1.05 | 1.02 | 1.03 | 1.05 | 1.13 | 1.09 |
| 1979 | 1.05 | 1.22 | 1.11 | 1.06 | 1.20 | 1.03 | 1.32 | 1.60 | 1.45 |
| 1980 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| 1981 | 1.74 | 2.17 | 1.95 | 1.51 | 1.63 | 1.57 | 0.91 | 0.71 | 0.81 |
| 1982 | 1.26 | 1.46 | 1.36 | 1.27 | 1.36 | 1.31 | 0.87 | 0.64 | 0.75 |
| 1983 | 1.37 | 1.66 | 1.52 | 1.47 | 1.60 | 1.54 | 0.85 | 0.58 | 0.73 |
| 1984 | 1.54 | 1.96 | 1.75 | 1.74 | 1.93 | 1.84 | 0.83 | 0.52 | 0.67 |
| 1985 | 1.36 | 1.69 | 1.52 | 1.69 | 1.83 | 1.76 | 0.83 | 0.51 | 0.67 |
| 1986 | 1.40 | 1.68 | 1.54 | 1.72 | 1.83 | 1.77 | 0.82 | 0.50 | 0.66 |
| 1987 | 1.38 | 1.48 | 1.43 | 1.52 | 1.55 | 1.54 | 0.82 | 0.50 | 0.66 |
| 1988 | 1.36 | 1.47 | 1.42 | 1.82 | 1.88 | 1.85 | 0.82 | 0.50 | 0.66 |
| X | 1.23 | 1.33 | 1.28 | 1.23 | 1.28 | 1.25 | 1.07 | 1.17 | 1.12 |
| S | 0.21 | 0.38 | 0.29 | 0.35 | 0.41 | 0.38 | 0.21 | 0.59 | 0.40 |
| | | | | | | | | | |

Source: author's calculation.

Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted

| Table 9 | Trend equatio | ns for real effe | ctive exc | change rates fi | or Kenya, | , Ethiopia | and the Sudan | l, 1980= | 100 | |
|--------------------------------|--------------------------------------|------------------|--------------|-----------------|--------------|---------------|---------------|----------------|------------------|----------------|
| Model | Coefficients | - | Kenya 2 | e | - | Ethiopia 2 | ĸ | - | Sudan 2 | ო |
| Linear | Intercept | 0.65 | 0.62 7 29 | 0.64 8.51 | 0.98 | 0.88 6.49 | 0.90 | 1.43 | 2.19 52.41 | 1.81 34.76 |
| | Slope | 0.06 | 0.07 | 0.06 | 0.02 | 0.05 | 0.03 | -0.04 | -0.10 | -0.07 |
| | ${ m R}^2$ | 10.16 0.85 | 8.83 0.81 | 9.46 0.93 | 3.61 0.35 | 87.21 0.44 | 3.37 0.39 | -15.32 0.93 | -15.06 · 0.92 | -15.14 0.93 |
| | S.E. | 0.14 | 0.18 | 0.16 | 0.19 | 0.15 | 0.17 | 0.06 | 0.16 | 0.11 |
| Exponential | Intercept | 0.46 | 0.42 | 0.44 | 0.88 | 0.61 | 0.70 | 1.55 | 2.51 | 2.03 |
| | Sinne | 3.74 | 2.69 | 3.16 0.20 | 3.28 | 3.65 0.20 | 7.01 | 23.45 | 13.37 0.05 | 16.00 2.11 |
| | 200 | 0.60 | 2.92 | 6.24 6.24 | 3.50 | 3.87 3.87 | 0.47 3.25 | -0.24 -7.86 | -0.05 -7.61 | -0.44 -7.68 |
| | R² | 0.70 | 0.65 | 0.68 | 0.38 | 0.44 | 0.35 | 0.77 | 0.76 | 0.76 |
| | S.Е | 0.19 | 0.24 | 0.22 | 0.16 | 0.28 | 0.18 | 0.10 | 0.29 | 0.20 |
| Source: autho Notes: 1 = E) | r's calculations. xport weighted; | 2 = Import w | eighted; | 3 = Trade v | veighted | | | | | |

40

INDICES OF EFFECTIVE EXCHANGE RATES

| Kenya | Ethiopia | Sudan | |
|-------|---|---|--|
| 9.65 | 2.93 | 1.56 | |
| 9.72 | 2.92 | 1.64 | |
| 9.45 | 2.61 | 1.93 | |
| 10.50 | 2.20 | 1.73 | |
| 6.69 | 2.49 | 1.66 | |
| 8.86 | 5.07 | 1.44 | |
| 9.36 | 3.76 | 1.77 | |
| 8.61 | 3.98 | 1.73 | |
| 8.50 | 4.05 | 1.58 | |
| 8.49 | 3.10 | 1.33 | |
| 8.18 | 2.80 | 1.06 | |
| 11.15 | 3.04 | 1.11 | |
| 15.15 | 3.34 | 0.65 | |
| 16.99 | 3.54 | 0.54 | |
| 16.99 | 4.13 | 0.39 | |
| 17.06 | 4.71 | 0.32 | |
| 15.86 | 3.85 | 0.16 | |
| 18.95 | 5.50 | 0.16 | |
| 20.88 | 5.70 | - | |
| | | | |
| | Kenya 9.65 9.72 9.45 10.50 6.69 8.86 9.36 8.61 8.50 8.49 8.18 11.15 15.15 16.99 17.06 15.86 18.95 20.88 | KenyaEthiopia9.652.939.722.929.452.6110.502.206.692.498.865.079.363.768.613.988.504.058.493.108.182.8011.153.0415.153.3416.994.1317.064.7115.863.8518.955.5020.885.70 | KenyaEthiopiaSudan9.652.931.569.722.921.649.452.611.9310.502.201.736.692.491.668.865.071.449.363.761.778.613.981.738.504.051.588.493.101.338.182.801.0611.153.041.1115.153.340.6516.993.540.5416.994.130.3917.064.710.3215.863.850.1618.955.500.1620.885.70- |

Table 10 Parallel exchange rates for Kenya, Ethiopia and the Sudan

Source: personal communications with Ibrahim Elbadawi. Note: values are expressed in terms of one US dollar.

| Ta | ble | 11 |
|----|-----|----|

11 Parallel effective exchange rates for Kenya, Ethiopia and the Sudan, 1970 = 100

| | | Kenya | | E | thiopia | | | Sudan | |
|------|------|-------|------|------|---------|------|------|-------|------|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 |
| 1970 | 1 00 | 1 00 | 1 00 | 1 00 | 1 00 | 1 00 | 1 00 | 1 00 | 1 00 |
| 1071 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| 1972 | 0.93 | 0.97 | 0.95 | 1.02 | 1.02 | 1.02 | 1.30 | 1 28 | 1 29 |
| 1973 | 0.83 | 0.94 | 0.88 | 1 14 | 1 12 | 1 13 | 1.00 | 1.18 | 1.21 |
| 1974 | 0.91 | 0.99 | 0.95 | 0.85 | 0.87 | 0.86 | 1.23 | 1.15 | 1.19 |
| 1975 | 1.53 | 1.35 | 1.44 | 1.01 | 1.02 | 1.02 | 1.08 | 1.08 | 1.08 |
| 1976 | 1 21 | 1 19 | 1 20 | 1.09 | 1.07 | 1.08 | 1.42 | 1.53 | 1.48 |
| 1977 | 1.27 | 1.25 | 1.26 | 1.06 | 1.04 | 1.05 | 1.40 | 1.41 | 1.41 |
| 1978 | 1.31 | 1.30 | 1.31 | 1.10 | 1.08 | 1.09 | 1.34 | 1.27 | 1.31 |
| 1979 | 1.07 | 1.17 | 1.12 | 1.11 | 1.10 | 1.10 | 1.13 | 1.02 | 1.08 |
| 1980 | 0.99 | 1.08 | 1.04 | 1.04 | 1.04 | 1.04 | 0.84 | 0.75 | 0.80 |
| 1981 | 1.04 | 1.08 | 1.06 | 1.25 | 1.22 | 1.23 | 1.07 | 1.44 | 1.26 |
| 1982 | 1.11 | 1.12 | 1.12 | 1.56 | 1.50 | 1.53 | 0.52 | 0.58 | 0.55 |
| 1983 | 1.15 | 1.12 | 1.14 | 1.64 | 1.56 | 1.60 | 0.43 | 0.51 | 0.47 |
| 1984 | 1.29 | 1.20 | 1.23 | 1.58 | 1.50 | 1.54 | 0.32 | 0.44 | 0.38 |
| 1985 | 1.44 | 1.31 | 1.38 | 1.70 | 1.62 | 1.66 | 0.26 | 0.31 | 0.29 |
| 1986 | 1.24 | 1.25 | 1.25 | 1.74 | 1.65 | 1.69 | 0.14 | 0.16 | 0.15 |
| 1987 | 1.67 | 1.63 | 1.65 | 1.20 | 2.05 | 2.13 | 0.14 | 0.14 | 0.14 |
| 1988 | 1.71 | 1.61 | 1.66 | 2.31 | 2.15 | 2.23 | | | |
| x | 1.19 | 1.19 | 1.19 | 1.34 | 1.30 | 1.32 | 0.85 | 0.90 | 0.89 |
| S | 0.25 | 0.19 | 0.22 | 0.42 | 0.37 | 0.40 | 0.45 | 0.45 | 0.45 |
| | | | | | | | | | |

Source: author's calculation, based on Table 10. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| | | | | | | | | | | | 1 |
|-------------|----------------|------|------------|------|-------|--------------|-------|-------------|------------|-------|---|
| Model | Coefficients | - | kenya 2 | ю | ш | thiopia 2 | ო | | sudan 2 | ю | |
| Linear | Intercept | 0.68 | 0.72 | 0.70 | 06.0 | 0.95 | 0.89 | 1.54 | 1.50 | 1.52 | |
| | | 6.71 | 8.04 | 7.33 | 9.70 | 13.83 | 11.61 | 11.31 | 9.82 | 10.68 | |
| | Slope | 0.07 | 0.06 | 0.06 | 0.03 | 0.02 | 0.03 | -0.07 | -0.06 | -0.07 | |
| | - | 7.44 | 7.34 | 7.34 | 3.65 | 4.02 | 3.89 | 5.47 | 4.12 | 4.99 | |
| | R ² | 0.75 | 0.75 | 0.75 | 0.41 | 0.45 | 0.44 | 0.63 | 0.52 | 0.58 | |
| | S.E. | 0.21 | 0.19 | 0.20 | 0.19 | 0.14 | 0.16 | 0.28 | 0.31 | 0.29 | |
| Exponential | Intercept | 0.59 | 0.63 | 0.61 | 0.82 | 0.88 | 0.82 | 1.54 | 1.50 | 1.52 | |
| - | | 2.89 | 3.56 | 3.12 | 6.0 | 8.69 | 7.23 | 11.31 | 9.82 | 10.68 | |
| | Slope | 0.36 | 0.32 | 0.34 | 0.18 | 0.15 | 0.16 | -0.07 | -0.06 | -0.07 | |
| | | 3.95 | 3.92 | 3.94 | 2.91 | 3.33 | 3.13 | -5.47 | -4.42 | 4.99 | |
| | \mathbb{R}^2 | 0.45 | 0.44 | 0.45 | 0.29 | 0.36 | 0.33 | 0.63 | 0.52 | 0.58 | |
| | S.E. | 0.32 | 0.28 | 0.30 | 0.21 | 0.16 | 0.18 | 0.28 | 0.31 | 0.29 | |
| | | | | | | | | | | | I |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| | | Kenya | | E | thiopia | | ; | Sudan | | |
|------|------|-------|------|------|---------|------|------|-------|------|---|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 | |
| 1970 | 0.88 | 0.95 | 0.92 | 0.95 | 0.93 | 0.94 | 1.00 | 1.00 | 1.00 | _ |
| 1971 | 0.91 | 0.98 | 0.95 | 0.96 | 0.94 | 0.95 | 1.11 | 1.06 | 1.08 | |
| 1972 | 0.90 | 0.97 | 0.94 | 0.91 | 0.90 | 0.91 | 1.26 | 1.30 | 1.28 | |
| 1973 | 1.04 | 1.08 | 1.06 | 0.85 | 0.86 | 0.86 | 1.10 | 1.11 | 1.11 | |
| 1974 | 0.80 | 0.83 | 0.82 | 0.92 | 0.92 | 0.92 | 0.98 | 0.96 | 0.97 | |
| 1975 | 0.94 | 0.98 | 0.96 | 1.40 | 1.30 | 1.35 | 0.79 | 0.92 | 0.86 | |
| 1976 | 1.01 | 1.03 | 1.02 | 1.16 | 1.12 | 1.14 | 1.15 | 1.50 | 1.33 | |
| 1977 | 1.00 | 1.01 | 1.01 | 1.22 | 1.18 | 1.20 | 1.04 | 1.31 | 1.68 | |
| 1978 | 1.07 | 1.06 | 1.06 | 1.27 | 1.23 | 1.25 | 0.88 | 1.04 | 0.96 | |
| 1979 | 1.09 | 1.06 | 1.07 | 1.08 | 1.08 | 1.08 | 0.62 | 0.71 | 0.66 | |
| 1980 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 0.40 | 0.46 | 0.43 | |
| 1981 | 1.17 | 1.18 | 1.18 | 1.02 | 1.01 | 1.02 | 0.52 | 0.92 | 0.72 | |
| 1982 | 1.45 | 1.45 | 1.45 | 1.07 | 1.05 | 1.06 | 0.22 | 0.34 | 0.28 | |
| 1983 | 1.47 | 1.52 | 1.50 | 1.08 | 1.05 | 1.06 | 0.14 | 0.22 | 0.18 | |
| 1984 | 1.38 | 1.46 | 1.42 | 1.17 | 1.12 | 1.15 | 0.08 | 0.14 | 0.11 | |
| 1985 | 1.56 | 1.58 | 1.57 | 1.32 | 1.25 | 1.29 | 0.05 | 0.07 | 0.06 | |
| 1986 | 1.67 | 1.62 | 1.64 | 1.26 | 1.17 | 1.19 | 0.02 | 0.03 | 0.02 | |
| 1987 | 2.14 | 2.05 | 2.12 | 1.59 | 1.53 | 1.56 | 0.02 | 0.02 | 0.02 | |
| 1988 | 2.25 | 2.13 | 2.20 | 1.60 | 1.53 | 1.56 | - | - | - | |
| x | 1.25 | 1.26 | 1.26 | 1.47 | 1.12 | 1.13 | 0.63 | 0.73 | 0.69 | |
| S | 0.44 | 0.38 | 0.40 | 0.22 | 0.19 | 0.20 | 0.45 | 0.49 | 0.47 | |

Table 13Parallel nominal effective exchange rates for Kenya, Ethiopia and the
Sudan

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| Table 14 | Trend equations for 1980=100 | parallel | market | nominal ef | fective exch | lange rat | es for Kenya, | Ethiopia a | and the | Sudan, |
|-------------|---------------------------------|----------|--------|------------|--------------|-----------|---------------|------------|---------|--------|
| | | | Kenya | | | Ethiopia | | | Sudan | |
| Model | Coefficients | - | ้ณ | ო | - | ึง | ო | - | 0 | ო |
| Linear | Intercept | 0.59 | 0.67 | 0.63 | 0.89 | 0.88 | 0.88 | 2.02 | 1.87 | 1.95 |
| | | 5.98 | 7.50 | 6.74 | 11.31 | 12.93 | 12.08 | 9.48 | 11.52 | 10.53 |
| | Slope | 0.07 | 0.06 | 0.06 | 0.03 | 0.02 | 0.02 | -0.08 | -0.08 | -0.08 |
| | | 7.59 | 7.50 | 7.59 | 3.74 | 3.98 | 3.86 | -4.21 | -5.60 | -4.89 |
| | \mathbb{R}^2 | 0.76 | 0.75 | 0.76 | 0.42 | 0.45 | 0.43 | 0.49 | 0.65 | 0.57 |
| | S.E. | 0.21 | 0.19 | 0.20 | 0.16 | 0.14 | 0.12 | 0.43 | 0.33 | 0.38 |
| Exponential | Intercept | 0.49 | 0.59 | 0.54 | 0.82 | 0.81 | 0.88 | 2.02 | 1.87 | 1.95 |
| | | 2.45 | 3.24 | 2.83 | 0.71 | 8.00 | 12.10 | 9.48 | 11.52 | 10.53 |
| | Slope | 0.37 | 0.33 | 0.35 | 0.16 | 0.15 | 0.02 | -0.08 | -0.08 | -0.08 |
| | | 4.13 | 3.99 | 4.07 | 3.07 | 3.26 | 3.86 | -4.21 | -5.60 | -4.89 |
| | \mathbb{R}^2 | 0.47 | 0.45 | 0.46 | 0.32 | 0.35 | 0.43 | 0.25 | 0.64 | 0.57 |
| | S.E. | 0.31 | 0.28 | 0.29 | 0.18 | 0.16 | 0.15 | 0.43 | 0.33 | 0.38 |

| | trade weighted. |
|--------------------------------|---------------------------------|
| | import weighted; 3 = |
| Source: author's calculations. | Notes: 1 = export weighted; 2 = |

| | 1370 | - 100 | | | | | | | |
|------|------|-------|------------|------|---------|------|------|-------|------|
| | | Kenya | , <u> </u> | E | thiopia | | | Sudan | |
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 |
| 1970 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| 1971 | 1.05 | 1.06 | 1.06 | 1.11 | 1.14 | 1.13 | 1.11 | 1.06 | 1.09 |
| 1972 | 1.10 | 1.12 | 1.11 | 1.08 | 1.20 | 1.14 | 1.26 | 1.29 | 1.27 |
| 1973 | 1.26 | 1.25 | 1.25 | 0.96 | 1.13 | 1.04 | 1.10 | 1.10 | 1.10 |
| 1974 | 0.93 | 0.96 | 0.94 | 1.08 | 1.25 | 1.17 | 0.99 | 0.97 | 0.98 |
| 1975 | 1.05 | 1.08 | 1.06 | 1.90 | 1.83 | 1.86 | 0.79 | 0.89 | 0.85 |
| 1976 | 1.13 | 1.16 | 1.15 | 1.31 | 1.43 | 1.37 | 1.15 | 1.44 | 1.29 |
| 1977 | 1.07 | 1.10 | 1.09 | 1.27 | 1.42 | 1.35 | 1.04 | 1.28 | 1.15 |
| 1978 | 1.07 | 1.11 | 1.10 | 1.23 | 1.38 | 1.30 | 0.87 | 1.00 | 0.93 |
| 1979 | 1.11 | 1.15 | 1.13 | 0.99 | 1.21 | 1.10 | 0.61 | 0.68 | 0.64 |
| 1980 | 1.07 | 1.12 | 1.10 | 1.02 | 1.27 | 1.15 | 0.39 | 0.44 | 0.42 |
| 1981 | 1.21 | 1.25 | 1.23 | 1.10 | 1.33 | 1.22 | 0.51 | 0.85 | 0.68 |
| 1982 | 1.24 | 1.25 | 1.25 | 1.20 | 1.39 | 1.29 | 0.22 | 0.31 | 0.27 |
| 1983 | 1.17 | 1.18 | 1.17 | 1.28 | 1.46 | 1.37 | 0.13 | 0.20 | 0.17 |
| 1984 | 1.09 | 1.10 | 1.10 | 1.36 | 1.45 | 1.41 | 0.07 | 0.13 | 0.10 |
| 1985 | 1.06 | 1.07 | 1.07 | 1.54 | 1.62 | 1.58 | 0.04 | 0.07 | 0.06 |
| 1986 | 1.01 | 1.02 | 1.02 | 1.35 | 1.55 | 1.45 | 0.02 | 0.03 | 0.02 |
| 1987 | 1.20 | 1.19 | 1.19 | 1.78 | 1.90 | 1.84 | 0.02 | 0.02 | 0.02 |
| 1988 | 1.37 | 1.36 | 1.36 | 1.77 | 1.84 | 1.80 | - | - | - |
| x | 1.14 | 1.13 | 1.12 | 1.28 | 1.41 | 1.35 | 0.67 | 0.71 | 0.67 |
| S | 0.11 | 0.10 | 0.10 | 0.28 | 0.25 | 0.26 | 0.46 | 0.48 | 0.47 |

Table 15Parallel real effective exchange rates for Kenya, Ethiopia and the Sudan,
1970 = 100

Source: author's calculation. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| Table 16 | Trend equations for I | oarallel m | larket rea | al effective exc | hange rat | es for Ke | nya, Ethiopia | t and the St | 197 197 | 70=100 |
|-------------|-----------------------|------------|------------|------------------|-----------|-----------|---------------|--------------|------------|--------|
| | | | Kenva | | | Ethiopia | | | Sudan | |
| Model | Coefficients | ⊷ | °0 | e | - | N | e | - | 5 | 93 |
| Linear | Intercept | 1.04 | 1.06 | 1.05 | 1.01 | 1.09 | 1.06 | 1.39 | 1.47 | 1.43 |
| | - | 22.31 | 24.08 | 23.27 | 8.70 | 12.81 | 10.52 | 16.85 | 11.16 | 13.60 |
| | Slope | 0.01 | 0.01 | 0.01 | 0.03 | 0.03 | 0.03 | -0.08 | -0.08 | -0.08 |
| | - | 1.98 | 1.93 | 1.97 | 2.69 | 4.24 | 3.38 | -10.45 | -6.40 | -8.11 |
| | R² | 0.14 | 0.13 | 0.14 | 0.26 | 0.49 | 0.37 | 0.86 | 0.70 | 0.79 |
| | S.E. | 0.10 | 0.10 | 0.09 | 0.24 | 0.18 | 0.20 | 0.17 | 0.28 | 0.21 |
| Exponential | Intercept | 1.01 | 1.02 | 1.02 | 0.93 | 0.97 | 0.95 | 1.54 | 1.56 | 1.55 |
| | - | 15.80 | 17.31 | 16.60 | 5.69 | 8.11 | 6.73 | 8.31 | 6.39 | 7.28 |
| | Slope | 0.05 | 0.05 | 0.05 | 0.17 | 0.21 | 0.19 | -0.45 | -0.41 | -0.48 |
| | - | 1.81 | 2.02 | 1.92 | 2.28 | 3.98 | 3.01 | -5.26 | -3.65 | -4.38 |
| | \mathbb{R}^2 | 0.11 | 0.15 | 0.13 | 0.19 | 0.45 | 0.31 | 0.61 | 0.42 | 0.52 |
| | S.E. | 0.10 | 0.10 | 0.10 | 0.25 | 0.18 | 0.22 | 0.28 | 0.38 | 0.38 |
| | | | | | | | | | | |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| | | Kenya | | E | Ethiopia | | | Sudan | |
|------|------|-------|------|------|----------|------|------|-------|------|
| Year | 1 | 2 | 3 | 1 | 2 | 3 | 1 | 2 | 3 |
| 1970 | 0.87 | 0.89 | 0.88 | 0.86 | 0.81 | 0.84 | 2.24 | 2.51 | 2.37 |
| 1971 | 0.93 | 0.95 | 0.94 | 0.97 | 0.92 | 0.95 | 2.38 | 2.76 | 2.57 |
| 1972 | 0.99 | 0.99 | 0.99 | 0.99 | 0.95 | 0.97 | 2.87 | 3.17 | 3.02 |
| 1973 | 1.15 | 1.13 | 1.14 | 0.91 | 0.90 | 0.90 | 2.47 | 2.77 | 2.61 |
| 1974 | 0.87 | 0.85 | 0.86 | 1.02 | 0.99 | 1.01 | 2.17 | 2.46 | 2.32 |
| 1975 | 0.96 | 0.96 | 0.96 | 1.64 | 1.50 | 1.57 | 2.01 | 2.01 | 2.01 |
| 1976 | 1.05 | 1.04 | 1.05 | 1.20 | 1.15 | 1.17 | 3.22 | 2.92 | 3.07 |
| 1977 | 1.01 | 0.99 | 1.00 | 1.14 | 1.10 | 1.12 | 2.82 | 2.63 | 2.73 |
| 1978 | 1.02 | 0.99 | 1.01 | 1.14 | 1.10 | 1.12 | 2.24 | 2.21 | 2.23 |
| 1979 | 1.06 | 1.03 | 1.05 | 0.96 | 0.95 | 0.96 | 1.53 | 1.55 | 1.54 |
| 1980 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 | 1.00 |
| 1981 | 1.10 | 1.11 | 1.05 | 1.05 | 1.05 | 1.05 | 1.91 | 1.35 | 1.63 |
| 1982 | 1.10 | 1.12 | 1.11 | 1.12 | 1.10 | 1.11 | 0.71 | 0.58 | 0.64 |
| 1983 | 1.01 | 1.05 | 1.03 | 1.18 | 1.16 | 1.17 | 0.46 | 0.35 | 0.40 |
| 1984 | 0.92 | 0.98 | 0.96 | 1.21 | 1.67 | 1.19 | 0.28 | 0.20 | 0.24 |
| 1985 | 0.93 | 0.96 | 0.95 | 1.37 | 1.30 | 1.33 | 0.16 | 0.12 | 0.14 |
| 1986 | 0.92 | 0.92 | 0.92 | 1.26 | 1.23 | 1.25 | 0.06 | 0.05 | 0.06 |
| 1987 | 1.11 | 1.06 | 1.09 | 1.61 | 1.53 | 1.57 | 0.05 | 0.05 | 0.05 |
| 1988 | 1.25 | 1.21 | 1.23 | 1.58 | 1.48 | 1.53 | - | - | - |
| X | 1.01 | 1.01 | 1.01 | 1.17 | 1.30 | 1.15 | 1.59 | 1.59 | 1.59 |
| S | 0.10 | 0.09 | 0.09 | 0.23 | 0.21 | 0.22 | 1.07 | 1.40 | 1.10 |
| | | | | | | | | | |

| Table 17 | Parallel real effective exchange rates for Kenya, Ethiopia and the Sudan, |
|----------|---|
| | 1980 = 100 |

Source: author's calculation.

Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

| 1980=100 |
|--|
| Ethiopia and the Sudan, |
| equations for parallel market real effective exchange rates for Keny |
| able 18 Trend |

| | | | | | I | | | | | |
|-------------|----------------|--------|---------|-------|--------|---------|-------|-------|--------|-------|
| | Coafficiante | - | Kenya | e c | ш - | thiopia | r | | Sudan | ď |
| | | - | u | 5 | - | J | 0 | - | L | , |
| Linear | Intercept | 0.95 | 0.95 | 0.95 | 0.91 | 0.88 | 0.90 | 3.24 | 3.49 | 3.36 |
| | | 21.03. | 24.02.2 | 2.67 | 10.25 | 11.90 | 11.00 | 11.91 | 6.67 | 14.16 |
| | Slope | 0.01 | 0.01 | 0.01 | 0.03 | 0.03 | 0.03 | -0.17 | -0.20 | -0.19 |
| | | 1.73 | 1.89 | 1.82 | 3.30 | 3.89 | 3.57 | 6.92 | -10.31 | -8.50 |
| | \mathbb{R}^2 | 0.09 | 0.13 | 0.11 | 0.35 | 0.44 | 0.40 | 0.73 | 0.86 | 0.81 |
| | S.E. | 0.09 | 0.08 | 0.08 | 0.19 | 0.15 | 0.17 | 0.65 | 0.43 | 0.49 |
| Exponential | Intercept | 0.92 | 0.91 | 3.47 | 0.78 | 0.80 | 0.91 | 3.47 | 3.87 | 3.67 |
| | | 15.15 | 11.33 | 16.34 | 8.57 | 7.55 | 7.01 | 6.69 | 8.25 | 7.48 |
| | Slope | 0.05 | 0.05 | 0.05 | 0.17 | 0.17 | 0.17 | -0.93 | -1.12 | -1.03 |
| | | 1.93 | 1.99 | 1.98 | 2.96 | 3.59 | 3.25 | -3.89 | -5.28 | -4.54 |
| | \mathbb{R}^2 | 0.13 | 0.14 | 0.14 | 3.30 | 0.40 | 0.35 | 0.45 | 0.61 | 0.54 |
| | S.E. | 0.09 | 0.08 | 0.09 | 0.19 | 0.16 | 0.18 | 0.79 | 0.72 | 0.75 |
| | | | | | | | | | | |

Source: author's calculations. Notes: 1 = export weighted; 2 = import weighted; 3 = trade weighted.

Figure 7 Parallel nomimal effective exchange rates for Kenya, 1970=100



Figure 8 Parallel nominal effective exchange rates for Ethiopia, 1970=100







X Conclusions

In this study we have attempted to ascertain whether the standard measures of nominal and real EERs can be applied within the African context by critically analyzing various indices; while the computations are not difficult the interpretations and comparisons are not easy and could be misleading. the contribution of this paper will thus be along these lines. This analysis shows that the measurements of the NEER indices can be readily applied to data collected form Africa. Our results for Ethiopia, Kenya and the Sudan showed that the NEER indices give some indication of the extent of overvaluation of the currency of these countries.

When it comes to measuring REER, the results are not as explicit as those for the NEER. The reason for this is that when the NEER is deflated by inflation differentials between the African countries under study and the corresponding partner countries, the result seems to be less reliable. this is especially so for Ethiopia and the Sudan, while the results for Kenya are still reliable. The measures of inflation of a reporting and partner country are the CPI; for developed countries the CPI is a fairly reliable measure, while for African countries it is not. Furthermore, the CPIs for Kenya, Ethiopia and the Sudan seem to be confined to urban centres. In estimating the parallel nominal and real exchange rates, the nominal rates gave very interesting results, especially for Ethiopia. This indicates that the parallel exchange rate may be a better estimate of the value of the Ethiopian currency and, to some extent, or the Sudanese and Kenyan currency. The indices are also related to key macroeconomic indicators in the three countries.

In general, the selected macroeconomic indicators of the three countries suggest a more realigned exchange rate may lead to higher volume of exports, lower import export ratio, higher ratio of price of tradables to non-tradables and lower budgetary deficit to GDP ratio.

Appendix 1

Distribution of export by partner countries, 1970 and 1980

| | Ethic | opia's exp | orts* | Bilateral exchan | ige rate | |
|----------------|-------|------------|---------|------------------|----------|---------|
| | 1970 | 1980 | 1980/70 | 1970 | 1980 | 1980/70 |
| United States | 0.77 | 0.33 | 0.43 | 2.50 | 2.07 | 0.826 |
| Japan | 0.05 | 0.11 | 2.20 | 0.007 | 0.010 | 1.429 |
| France | 0.02 | 0.12 | 6.00 | 0.318 | 0.459 | 1.443 |
| Germany | 0.08 | 0.17 | 2.13 | 0.685 | 1.056 | 1.541 |
| Italy | 0.06 | 0.19 | 3.17 | 0.004 | 0.002 | 0.500 |
| Netherlands | 0.01 | 0.4 | 4.00 | 0.735 | 0.972 | 1.322 |
| United Kingdom | 0.13 | 0.04 | 1.33 | 1.046 | 0.866 | 0.828 |
| | Ker | iya's exp | orts* | Bilateral exchan | ge rate* | |
| United States | 0.18 | 0.10 | 0.56 | 7.14 | 7.57 | 3.54 |
| Japan | 0.03 | 0.03 | 1.00 | 0.01 | 0.04 | 4.00 |
| France | 0.02 | 0.04 | 2.00 | 0.33 | 1.68 | 5.09 |
| Germany | 0.19 | 0.31 | 1.63 | 0.59 | 3.86 | 6.54 |
| Italy | 0.04 | 0.12 | 3.00 | 0.003 | 0.008 | 2.67 |
| Netherlands | 0.10 | 0.10 | 1.00 | 0.63 | 3.55 | 5.63 |
| United Kingdom | 0.44 | 0.30 | 0.68 | 0.90 | 3.17 | 3.52 |
| | Su | dan' expo | orts* | Bilateral exchan | ge rate* | |
| United States | 0.09 | 0.03 | 0.33 | 2.87 | 2.00 | 0.70 |
| Japan | 0.24 | 0.12 | 0.50 | 0.008 | 0.01 | 1.25 |
| France | 0.07 | 0.04 | 0.57 | 0.440 | 0.44 | 1.00 |
| Germany | 0.14 | 0.07 | 0.50 | 0.79 | 1.02 | 1.29 |
| Italy | 0.24 | 0.17 | 0.71 | 0.005 | 2.22 | 4.44 |
| Netherlands | 0.08 | 0.03 | 0.38 | 0.84 | 0.94 | 1.12 |
| United Kinadom | 0.14 | 0.54 | 8 86 | 1 20 | 0.84 | 0.70 |

Source: author's calculation

Notes: *standardized; ** reporting over partner country

Appendix 2

Distribution of import by partner countries, 1970 and 1980

| | Ethiopia's imports* | | | Bilateral exchange rate | | | |
|----------------|---------------------|--------|---------|---------------------------|--|--|--|
| | 1970 | 1980 (| 1980/70 | 1970 1980 1980/70 | | | |
| United States | 0.13 | 0.19 | 1.46 | 0.40 0.48 1.20 | | | |
| Japan | 0.22 | 0.21 | 0.95 | 143.20 98.07 0.68 | | | |
| France | 0.04 | 0.03 | 0.75 | 3.14 2.18 0.69 | | | |
| Germany | 0.21 | 0.20 | 0.95 | 1.46 0.95 0.65 | | | |
| Italy | 0.25 | 0.20 | 0.80 | 250.00 500.00 2.00 | | | |
| Netherlands | 0.04 | 0.04 | 1.00 | 1.36 1.03 0.76 | | | |
| United Kingdom | 0.11 | 0.13 | 1.18 | 1.15 1.21 1.05 | | | |
| | Kenya's imports* | | | Bilateral exchange rate* | | | |
| United States | 0.13 | 0.13 | | 0.47 0.13 0.28 | | | |
| Japan | 0.16 | 0.19 | 1.19 | 100.00 25.00 0.25 | | | |
| France | 0.06 | 0.03 | 0.50 | 3.03 0.60 0.20 | | | |
| Germany | 0.12 | 0.17 | 1.42 | 1.69 0.26 0.15 | | | |
| Italy | 0.07 | 0.08 | 1.14 | 333.33 125.00 0.37 | | | |
| Netherlands | 0.04 | 0.06 | 1.50 | 1.59 0.28 0.18 | | | |
| United Kingdom | 0.42 | 0.34 | 0.81 | 1.11 0.32 0.28 | | | |
| | Sudan' imports* | | | Bilateral exchange rate** | | | |
| United States | 0.06 | 0.14 | 2.33 | 0.35 0.50 1.43 | | | |
| Japan | 0.12 | 0.10 | 0.91 | 125.00 100.00 0.80 | | | |
| France | 0.02 | 0.19 | 9.50 | 2.27 2.27 1.00 | | | |
| Germany | 0,13 | 0.17 | 1.31 | 1.27 0.98 0.78 | | | |
| Italy | 0.04 | 0.07 | 1.75 | 200.00 0.45 0.002 | | | |
| Netherlands | 0.04 | 0.09 | 2.25 | 1,19 1,06 0.89 | | | |
| United Kingdom | 0.60 | 0.24 | 0.40 | 0.83 1.19 1.43 | | | |

Source: author's calculation

Notes: *standardized; ** reporting over partner country

Appendix 3

| | | | | Ethiopia | | | | |
|----------|---------|---------|-------------------|----------|----------------|----------------|-----------------|-------------|
| | *Export | *Import | Import/ Export | | Export/ GDP | Import/ GDP | Deficit/ GDP | CPI |
| 1975 | 478.3 | 613.1 | 1.28 | | 0.19 | 0.25 | ····· | · · · · · · |
| 1980 | 879.3 | 1494.7 | 1.70 | | 0.21 | 0.36 | 0.01 | 100 |
| 1985 | 332.9 | 840.5 | 2.53 | | 0.07 | 0.18 | 0.05 | 144 |
| 1986 | 477.1 | 932.6 | 1.95 | | 0.09 | 0.18 | 0.09 | 130 |
| 1987 | 385.2 | 932.7 | 2.42 | | 0.07 | 0.17 | 0.07 | 127 |
| 1988 | 429.3 | 956.7 | 2.23 | | 0.07 | 0.17 | 0.07 | 136 |
| | | | | Kenya | | | | |
| 1975 | 572.4 | 814.2 | 1.40 | | 0.25 | 0.35 | 0.03 | |
| 1980 | 1242.6 | 2396.4 | 1.93 | | 0.17 | 0.33 | 0.02 | 100 |
| 1985 | 943.2 | 1269.8 | 1.35 | | 0.15 | 0.20 | 0.03 | 187 |
| 1986 | 1170.2 | 12454.6 | 1.24 | | 0.16 | 0.20 | 0.05 | 194 |
| 1987 | 908.7 | 1622.6 | 1.78 | | 0.11 | 0.20 | 0.07 | 204 |
| 1988 | 1017.5 | 1802.2 | 1.77 | | 0.12 | 0.21 | 0.04 | 221 |
| | | | | Sudan | | | | |
| 1975 | 407.7 | 748.3 | 1.83 | | 0.21 | 0.40 | 0.01 | |
| 1980 | 689.4 | 1127.4 | 1.64 | | 0.25 | 0.49 | 0.02 | 100 |
| 1985 | 442.2 | 579.0 | 1.30 | | 0.07 | 0.09 | | 398 |
| 1986 | 326.8 | 633.9 | 1.93 | | 0.05 | 0.12 | | 516 |
| 1987 | 265.7 | 696.6 | 2.62 | | 0.05 | 0.13 | | 650 |
| 1988 | 427.0 | 948.5 | 2.20 | | 0.06 | 0.13 | | 965 |
| | | | | | | | | |

Some indicators of macroeconomic performance for Ethiopia, Kenya and the Sudan

Note: *Merchandise exports and imports (in U.S. dollars)

Notes

- 1. Some common traditional methods are provided in the paper by Rhomberg (1976). They are similar to standard price and quantity indexes. On the other hand, the modern method which is commonly referred to as multilateral effective exchange rate (MERM) uses a partial equilibrium analysis. Empirical studies are available in several publications; for example Belanguer (1976) has developed a MERM for Zaire.
- 2. The real effective exchange rate (REER) may help us to measure the relative price of tradables and non-tradables; however, such interpretation depends on several critical assumptions.
- 3. After calculating the indices for the period under consideration, a simple linear $(Y_i = \beta_0 + \beta_1 X_i + \text{epsilon}_i)$ and nonlinear $(\ln Y_i = \ln \beta_i + \beta_i \ln X_i + \ln \text{epsilon}_i)$ are fitted using ordinary least squares (OLS). The aim here is to see if there is periodic fluctuation in the indices. A low R⁻² and a high standard error of coefficient (especially the seope) may indicate wide and unpredictable fluctuations and vice versa.

References

- Allen, R.G.D. (1975) Index Numbers in Theory and Practice, Chicago.
- Artus, J.R. (1974) "The behaviour of export prices for manufacturers", IMF Staff Papers, 21: 583-604.
- Artus, J.R. and Rhomber, R.R. (1973) "A multilateral exchange rate model", IMF Staff Papers, 20: 591-611.
- Balassa, B. (1975) "Reforming the Systems of Incentives in Developing Countries", World Development, 3: 365-81.
 (1982) "Structural adjustment policies in developing countries", World

Development, 10: 114-36.

(1987) "Effects of exchange rate changes in developing countries", Indian Journal of Economics.

- Bautista, R.M. (1981) "Exchange rate changes and LDC performance under generalized currency floating", *Weltwirstschafillches Archive*, 117: 443-68.
- Belanger, G. (1976) "An indicator of effective exchange rates for primary producing countries", IMF Staff Papers, 23: 113-36.
- Bhagwat, A. and Y. Onitsaka (1974) "Export-import responses to devaluation: experiences of the non-industrial countries in the 1960s", IMF Staff Papers, 21: 416-62.
- Black, S.W. (1976a) "Multilateral and bilateral measures of effective exchange rates in a world model of traded goods", JPE, 84: 615-21.
 (1976b) "Exchange policies for less developed countries in a world of floating rates", Essays in International Finance, No. 119, International Finance Section, Princeton University, Princeton.
- Bond, M.E. (1985) "Export demand and supply for groups of non-oil developing countries", IMF Staff Papers, 32: 56-77.
- Brodsky, D.A. (1980) "Calculating effective exchange rates: some practical implications", General Agreement on Tariff and Trade (GATT), mimeo, London.

- Coes, O.V. (1978) "The crawling peg and exchange rate uncertainty", a paper presented at a conference on The Crawling Peg: Past Performance and Future Prospects, Rio de Janiero 1976.
- Crockett, A.D. (1984) "Stabilization policies in developing countries: some policy considerations, IMF Staff Papers, 31: ?
- Crockett, A.D. and S.M. Nsouli (1977) "Exchange rate policies for developing countries", Journal of Development Studies, 13: 124-43.
- Diaz-Alejandro and F. Carlos (1965) Exchange Rate Devaluation in a Semi Industrialized Country, Cambridge, MA: MIT Press.
- Donovan, D.J. (1982) "Macroeconomic performance and adjustment under fund supported program: the experience of the seventies", IMF Staff Papers, 29: 171-203.
- Drechsler, L. (1973) "Weighting of index numbers in international comparisons", *Review of Income and Wealth*, 19: 17-34.
- Edwards, S. (1986) "Are devaluations contractionary?", *Review of Economics and Statistics*, 68: 501-08.
 (1988) "Exchange rate misalignment in developing countries", World Bank Occasional Paper, No. 20.
- Friedman, S. (1953) "The case for flexible exchange rate", in *Essays in Positive Economics*, Chicago: University of Chicago Press.
- Goldstein, M. (1977) "New measures of prices and productivity for tradable and non-tradable goods", *Review of Income and Wealth*, 25: 413-27.
- Goldstein, M., M.S. Khan and L.H. Officer (1980) "Prices of tradable and non-tradable goods in the demand for total imports", *Review of Economics* and Statistics, 62: 453-87.
- Gulhati, R., S. Bose and V. Atukorala (1985) "Exchange rate policies in Eastern and Southern Africa, 1965-83", World Bank Staff Working Papers, No. 720, Washington DC.
- Helleiner, G. (1989) "Structural adjustment and long-term development in sub-Saharan Africa", mimeo.
- Hirsch, F. and I. Higgins (1970) "An indicator of effective exchange rates", IMF Staff Papers, 17: 453-87.
- Isard, P. (1978) "Exchange rate determination: a survey of popular views and recent models", *Princeton Studies in International Finance*, No. 42, International Finance Section, Princeton University, Princeton, NJ.
- Khan, M.S. (1974) "Import and export demand in developing countries", IMF Staff Papers, 21: 679-93.

- Krugman, P. and L. Taylor (1978) "Contractionary effects on devaluations", Journal of International Economics, 12.
- Lipschitz, L. (1979) "Exchange rate policy for a small developing country, and the selection of appropriate standard", IMF Staff Papers, 26: 423-49.
- Maciejewski, E.M. (1983) "Real effective exchange rate indices", IMF Staff Papers, 30: 491-541.
- Morgan, T. and A. Davis (1982) "The concomitants of exchange rate depreciation; less developed countries, 1971-73", Economic Development and Cultural Change, 32: 101-27.
- Rana, P.B. (1981) "Exchange rate risk under generalized floating: eight Asian countries", Journal of International Economics, 11: 459-66.
- Rhomberg, R.R. (1976) "Indices of effective exchange rates", IMF Staff Papers, 23: 88-112.
- Robinson, W., T.R. Webb, and M.N. Townsend (1979) "The influences of exchange rate changes on prices: a study of 18 industrial countries", *Economica*, 46: 27-50.
- Ruggles, R. (1967) Price Indexes and International Price Comparisons; in Ten Economic Studies in the Tradition of Irving Fisher, Eilliam Fellner et al. (eds) New York: McGraw.
- Sato, K. (1976) "The Meaning and Measurement of the Real Value Added Index", *Review of Economics and Statistics*, 58: 432-42.
- Underwood, T.G. (1973) "Analysis of Proposals for Using Objective Indicators as a Guide to Exchange Rate Changes", *IMF Staff papers*, 20: 100-17.
- Williamson, T. (1983) The Open Economy and the World Economy, New York: Basic Books.
- Williamson, J. (1981) "A Survey of the Literature on the Optimal Peg", Journal of Development Economics, 11: 390-61.
AFRICAN ECONOMIC RESEARCH CONSORTIUM



P.O. BOX 62882 NAIROBI, KENYA

TELEPHONE (254-2) 228057 225234 215898 212359 332438 225087

TELEX 22480

FAX (254-2) 219308

The principal objective of the African Economic Research Consortium (AERC), established in August 1988, is to strengthen local capacity for conducting independent, rigorous inquiry into problems pertinent to the management of economies in Sub-Saharan Africa.

In response to special needs of the region, AERC has adopted a flexible approach to improve the technical skills of local researchers, allow for regional determination of research priorities, strengthen national institutions concerned with economic policy research, and facilitate closer ties between researchers and policy makers.

Since its establishment, AERC has been supported by private foundations, bilateral aid agencies and international organizations.

SPECIAL PAPERS contain the findings of commissioned studies in furtherance of AERC's programmes for research, training and capacity building.

RESEARCH PAPERS contain the edited and externally reviewed results of research financed by the AERC.

It is AERC's policy that authors of Special and Research Papers are free to use material contained therein in other publications. Views expressed in the Special and Research Papers are those of the authors alone and should not be attributed to the AERC's sponsoring Members, Advisory Committee, or Secretariat.

Further information concerning the AERC and additional copies of Special and Research Papers can be obtained by writing to: African Economic Research Consortium, P.O. Box 62882, Nairobi, Kenya.